

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dick, Christian D.; Schmeling, Maik; Schrimpf, Andreas

Working Paper Macro expectations, aggregate uncertainty, and expected term premia

ZEW Discussion Papers, No. 10-064

Provided in Cooperation with: ZEW - Leibniz Centre for European Economic Research

Suggested Citation: Dick, Christian D.; Schmeling, Maik; Schrimpf, Andreas (2010) : Macro expectations, aggregate uncertainty, and expected term premia, ZEW Discussion Papers, No. 10-064, Zentrum für Europäische Wirtschaftsforschung (ZEW), Mannheim

This Version is available at: https://hdl.handle.net/10419/41427

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Discussion Paper No. 10-064

Macro Expectations, Aggregate Uncertainty, and Expected Term Premia

Christian D. Dick, Maik Schmeling, and Andreas Schrimpf

ZEW

Zentrum für Europäische Wirtschaftsforschung GmbH

Centre for European Economic Research Discussion Paper No. 10-064

Macro Expectations, Aggregate Uncertainty, and Expected Term Premia

Christian D. Dick, Maik Schmeling, and Andreas Schrimpf

Download this ZEW Discussion Paper from our ftp server: ftp://ftp.zew.de/pub/zew-docs/dp/dp10064.pdf

Die Discussion Papers dienen einer möglichst schnellen Verbreitung von neueren Forschungsarbeiten des ZEW. Die Beiträge liegen in alleiniger Verantwortung der Autoren und stellen nicht notwendigerweise die Meinung des ZEW dar.

Discussion Papers are intended to make results of ZEW research promptly available to other economists in order to encourage discussion and suggestions for revisions. The authors are solely responsible for the contents which do not necessarily represent the opinion of the ZEW.

Non-technical summary

The empirical term structure literature shows that long-term interest rates are not merely a combination of expected short-term interest rates and a constant risk premium, as the so-called Expectation Hypothesis would suggest. Rather, the risk premium for holding long-term bonds varies over time and is predictable to a certain extent as shown by a multitude of studies. The economic reasons behind the movement of these risk premia are thus of special interest for monetary policy seeking to influence long-term market interest rates through a short-term key interest rate.

Our study contributes to a better understanding of risk premia in bond markets by explicitly analyzing expectations about bond risk premia. This analysis is based on individual data from the Survey of Professional Forecasters (SPF), conducted by the Federal Reserve Bank Philadelphia among forecasters from banks, financial institutions, and research institutions in the United States.

We empirically show that expectations about risk premia are significantly influenced by expectations about real macroeconomic activity (such as GDP), while nominal factors (such as inflation expectations) play a minor role. The uncertainty of forecasters about future business cycle movements also has a pronounced effect on risk premium expectations.

Our results indicate that expected risk premia are positively correlated with expectations about GDP and inflation. Higher uncertainty with respect to macroeconomic variables also increases expected risk premia. We also examine how the shape of the yield curve is related to expected risk premia. For instance, the curvature of the yield curve can be explained by information which is also captured by the expected change in risk premia of forecasters in the SPF.

Finally, we show that the expected changes in risk premia actually do forecast bond excess returns. This underlines that our proxy for risk premium expectations is indeed informative for developments in bond markets.

Nicht-technische Zusammenfassung

Die empirische Literatur zur Zinsstruktur zeigt, dass sich langfristige Zinsen nicht - wie dies die sogenannte Erwartungshypothese nahelegt - aus einer Kombination aus den für die Zukunft erwarteten kurzfristigen Zinsen und einer konstanten Risikoprämie zusammensetzen. Vielmehr ist die Risikoprämie für das Halten langfristiger Anleihen über die Zeit variabel und prognostizierbar. Für die Geldpolitik, die mit dem kurzfristigen Leitzins auch langfristige Marktzinsen beeinflussen möchte, sind die Gründe für die Bewegungen dieser Risikoprämien daher besonders interessant.

Unsere Studie trägt zu einem besseren Verständnis der Risikoprämien auf Anleihemärkten bei, indem sie explizit die *Erwartungen* hinsichtlich der Risikoprämien auf Zinsmärkten untersucht. Dies geschieht auf der Grundlage von umfragebasierten Individualdaten aus dem Survey of Professional Forecasters (SPF), die vierteljährlich von der Federal Reserve Bank Philadelphia unter Prognostikern aus Banken, Finanzunternehmen und Forschungsinstituten in den USA erhoben werden.

Wir zeigen empirisch, dass die Erwartungen bei Risikoprämien signifikant von Erwartungen hinsichtlich der realen makroökonomischen Entwicklungen (z.B. die Entwicklung des BIP) beeinflusst werden, während nominale Größen (z.B. Inflationserwartungen) eine im Vergleich geringere Rolle spielen. Auch die Unsicherheit, die die Prognostiker in Bezug auf die Einschätzung der künftigen makroökonomischen Entwicklung haben, wirkt sich auf die Erwartungen hinsichtlich der weiteren Entwicklung der Risikoprämien aus.

Nach unseren Ergebnissen stehen erwartete Risikoprämien in einem positiven Zusammenhang mit erwarteter wirtschaftlicher Dynamik und Inflationserwartungen. Auch die Unsicherheit über die Entwicklung wirtschaftlicher Größen führt dazu, dass mit einem Anstieg der Risikoprämien gerechnet wird. Weiterhin untersuchen wir, wie die Form der Zinsstrukturkurve mit den erwarteten Risikoprämien zusammenhängt. So wird offenbar insbesondere die Krümmung der Zinsstrukturkurve von Informationen beeinflusst, die sich auch aus den von uns gemessenen Erwartungen hinsichtlich der Veränderungen der Risikoprämien ergeben.

Wir weisen außerdem nach, dass die erwarteten Veränderungen der Risikoprämien Erklärungskraft für am Markt beobachtbare Überrenditen von langfristigen Anleihen haben. Dies unterstreicht, dass die von uns verwendeten Erwartungen tatsächlich das Geschehen an den Anleihemärkten verstehen helfen.

MACRO EXPECTATIONS, AGGREGATE UNCERTAINTY, AND EXPECTED TERM PREMIA *

Christian D. Dick[‡] Maik Schmeling^{**} Andreas Schrimpf[†]

August 27, 2010

Abstract

Based on individual expectations from the Survey of Professional Forecasters, we construct a real-time proxy for expected term premium changes on long-term bonds. We empirically investigate the relation of these bond term premium expectations with expectations about key macroeconomic variables as well as aggregate macroeconomic uncertainty at the level of individual forecasters. We find that expected term premia are (i) time-varying and reasonably persistent, (ii) strongly related to expectations about future output growth, and (iii) positively affected by uncertainty about future output growth and inflation rates. Expectations about real macroeconomic variables seem to matter more than expectations about nominal factors. Additional findings on term structure factors suggest that the level and slope factor capture information related to subjective term premium expectations themselves. Finally, an aggregate measure of forecasters' term premium expectations has predictive power for bond excess returns over horizons of up to one year.

JEL-Classification: E43, E44, G12

Keywords: Bond Yields, Expectations Hypothesis, Time-varying Risk Premia, Term Premia, Aggregate Uncertainty

^{*}We would like to thank Paul Carillo, Marcus Kappler, Kajal Lahiri, Juan Pablo Medina, Hashem Pesaran, Peter Norman Sørensen, Xuguang Sheng, as well as participants at the ZEW workshop on "Expectations and Expectation Formation in Financial Markets" (Mannheim, 2010) and the conference "Recent Developments in Macroeconomics" (Mannheim, 2010), and the annual conference of the European Economic Association (Glasgow, 2010) for helpful comments. Schmeling gratefully acknowledges support by the German Science Foundation (DFG) and Schrimpf appreciates financial support from the Center for Research in Econometric Analysis of Time Series, CREATES, funded by The Danish National Research Foundation.

[‡]Centre for European Economic Research (ZEW) Mannheim, P.O. Box 10 34 43, 68034 Mannheim, Germany, e-mail: dick@zew.de

^{**}Department of Economics, Leibniz Universität Hannover, Königsworther Platz 1, 30167 Hannover, Germany, e-mail: schmeling@gif.uni-hannover.de.

[†]Aarhus University and CREATES, School of Economics and Management, Bartholins Allé 10, DK-8000 Aarhus C, Denmark, e-mail: aschrimpf@creates.au.dk.

1 Introduction

Using panel data from the Survey of Professional Forecasters (SPF), we construct a simple proxy of forecaster-specific expectations about future term premium changes. These term premium expectations are not estimated, are available in real-time, clearly time-varying, and reasonably persistent. We then employ a dynamic panel regression framework to investigate macro determinants of these term premium expectations on the level of individual forecasters. Our results indicate that individual term premium expectations are most strongly influenced by *expectations* about real GDP growth and a measure of aggregate uncertainty about future real macro conditions. Inflation expectations and aggregate inflation uncertainty are also important, but are dominated by real factors. Finally, an aggregate measure of term premium expectations across forecasters has predictive power for future bond excess returns over forecast horizons of up to one year.

There is ample evidence that the expectations hypothesis of the term structure of interest rates does not hold empirically and that investors tend to demand a positive compensation for holding long-term bonds. These term premia – or bond risk premia – compensate investors for higher risk and drive a wedge between short rates controlled by the central bank and longer-maturity rates. Since the latter are crucial for spending and investment decisions in the economy, term premia are relevant in many branches of macroeconomics and finance. The literature has convincingly demonstrated that term premia are time-varying (see e.g. Cochrane and Piazzesi, 2005; Ludvigson and Ng, 2009) and at least partly driven by the state of the business cycle. As this adds complexity to the conduct of monetary policy as well as investment and borrowing decisions of the private and public sector, an active field of research is devoted to a better understanding of time-varying risk premia in bond markets (see e.g. Ang and Piazzesi, 2003; Rudebusch, 2010; Wright, 2009). As Gürkaynak and Wright (2010) point out, central banks could even target risk premia instead of short term interest rates (e.g., if those are at the zero bound) to bring long yields down.

This paper contributes to the strand of literature linking macroeconomic information to bond

yields and bond risk premia. While existing studies in this literature typically investigate aggregate term premium estimates, we focus on survey-based term premium expectations on the level of *individual* forecasters. This approach has two advantages: First, the reliance on survey information allows us to focus on the role of forward-looking macro *expectations* to understand term premia, whereas most earlier papers focus on the impact of *current* macro conditions.¹ Second, the forecasters in our panel naturally differ in their expectations about future macro conditions and term premia. We can exploit this cross-sectional variation to obtain powerful tests when analyzing determinants of bond risk premia. Unlike in aggregate data where cross-sectional differences in expectations are washed out,² individual data allow us to identify stable relationships between macro expectations and term premia in the cross-section of forecasters. The explicit focus on forward-looking macro variables and the use of individual real-time expectations is a novel aspect of our analysis compared to the existing literature.

Also, relative to pure time-series analyses of bond risk premia where future bond returns are regressed on current macro variables (e.g. Ludvigson and Ng, 2009) or other bond predictors (e.g. Cochrane and Piazzesi, 2005), our survey approach has the advantage that we are using *directly observable* term premium expectations (available in real-time) and do not have to estimate the expected return from noisy actual return data. This facilitates detecting potential links between macro variables and bond risk premia that may otherwise be overlooked.

Our main interest lies in the relation of term premia with expected future key macro business cycle variables - output growth and inflation - and aggregate uncertainty about these macro factors. Thus, in our benchmark specification, we regress individual term premium expectations on individual expectations about future real GDP growth and inflation (and instrument for these contemporaneous macro expectations) and measures of aggregate GDP and inflation uncertainty

¹One exception is Chun (2009) who studies the impact of macro expectations on the yield curve in the context of anticipated monetary policy. However, Chun deals with aggregate forecasts and does not study a panel of forecasters.

²For instance, consider the extreme case of only two investors where one forecaster expects a rise in the inflation rate of +2% and rising term premium of 1% and the other expects a decline in the inflation rate of -2% and a declining term premium of -1%. A panel regression will easily identify a positive relation between inflation expectations and term premia, whereas an aggregate analysis, relying on cross-sectional averages, will have little power to detect a significant relation since average inflation and term premium expectations will be equal to zero.

while controlling for lagged individual term premium expectations and further variables. As noted above, we find that nominal factors (expected inflation and inflation uncertainty) are important for term premia, but that real factors (expected output growth and uncertainty about future output growth) matter more and dominate the nominal factors. The relations are such that higher expectations about output growth and inflation raise expectations about future term premia. Likewise, higher aggregate uncertainty also raises term premium expectations.

Our results confirm findings from earlier papers which show that output growth and/or inflation matters for risk premia (Ang and Piazzesi, 2003; Bikbov and Chernov, 2008; Diebold, Rudebusch, and Aruoba, 2006; Chun, 2009; Ludvigson and Ng, 2009; Wright, 2009), and that macro uncertainty is important (Söderlind, 2009; Wright, 2009). Our findings differ from parts of the existing literature as we find that real factors clearly dominate nominal factors, especially when it comes to measures of economic uncertainty. This result may appear surprising at first sight, but it is robust to a variety of robustness tests and holds in the presence of real-time macro factors as well. Furthermore, we investigate the impact of aggregate risk (we mainly use realized bond market volatilities, credit spreads, and recession dummies to proxy for risk) and show that risk has the opposite effect on term premium expectations as uncertainty. Next, we investigate the relation of our term premium expectations with classic yield curve factors (level, slope, and curvature) as well as the Cochrane and Piazzesi (2005) return forecasting factor. We find that the level and slope of the yield curve seem to capture effects similar to our measures of aggregate uncertainty, which sheds some light on the economic forces underlying these two yield factors. Curvature, in turn, seems to be related to forecasters' term premium expectations themselves, which lines up with findings in Cochrane and Piazzesi (2008).

Finally, we test whether our proxy for term premium expectations is related to future bond excess returns. To this end, we run predictive regressions of bond returns on aggregate term premium expectations in the spirit of Cochrane and Piazzesi (2005) and Ludvigson and Ng (2009). We find that our real-time proxy for term premium expectations forecasts future bond excess returns with predictive R^2 s of up to 23% at an annual forecast horizon. This is quite remarkable in our view, given that our term premium proxy is a non-estimated variable (i.e. it is free of potential look-ahead or errors-in-variables problems) and is readily available in real-time. Furthermore, results from our forecasting exercise are in line with the view that our factor proxies for *future changes in risk premia* and not the current level of risk premia as in, e.g., Cochrane and Piazzesi (2005) or Ludvigson and Ng (2009), so that our factor differs from earlier proxies in the bond literature.

The paper proceeds as follows. The next section selectively reviews related literature, Section 3 describes the construction of our proxy for term premium expectations, Section 4 details the data and our panel regression framework, Section 5 presents empirical results, and Section 6 concludes. An appendix to this paper contains details on some of our data, variable construction, and our econometric approach and panel regression settings. A separate web appendix contains additional robustness analyses.

2 Related Literature

The Expectation Hypothesis (EH) of interest rates has been serving as a classical point of reference in economics and finance for decades. In its most basic form, it implies that bond risk premia (term premia) are constant over time.³ However, failures of this concept have been documented for more than 20 years. Early references include Fama and Bliss (1987) and Campbell and Shiller (1991) who show that the difference between forward rates and spot rates or the term spread, respectively, forecast bond (excess) returns. Taking account of this predictability, modern economic models understand risk premia in bond markets to be, in fact, time-varying. Due to the importance of term premia for economics and finance, there is now a vast literature on this topic. Hence, we do not attempt to survey the whole field but rather focus on a few selected studies which investigate the link between macro factors and bond yields. We also pay special attention to the use of survey

³There are different ways of stating the EH as well as its implications (cf. Cochrane, 2005). Here, we refer to the EH as the proposition that there are no time-varying term premia and that holding period excess returns on long-term bonds are not predictable. For a more precise statement see Eq. (1) in Sec. 3.

data for term premium modeling. For a more comprehensive literature overview, see e.g. Diebold, Piazzesi, and Rudebusch (2005), Kim (2009), or Gürkaynak and Wright (2010) for a survey on the link between the term structure and macroeconomics.

One approach of linking bond risk premia to macro factors is to run predictive regressions of future bond (excess) returns on current macro factors.⁴ This is done, e.g., by Ludvigson and Ng (2009) who extract macroeconomic factors from a large data set and find that bond returns are highly predictable with predictive R^2 s of up to 26% for U.S. bonds, indicating that term premia are clearly time-varying. They find that a real output factor is an important driver of bond excess returns. A related approach is to run predictive regressions on bond-related variables to compute forecasting factors, and relate these to the business cycle in a second step. For example, Cochrane and Piazzesi (2005) construct a factor based on a linear combination of forward rates for bonds with different maturities to forecast bond excess returns. These authors also show that their factor is related to the business cycle (see also Koijen, Lustig, and Van Nieuwerburgh, 2010, for a related analysis).

The link between yield curve and macro variables is also studied within macro-finance models that directly model macro factors together with yield curve models. For example, Ang and Piazzesi (2003) find that output shocks are an important driver of curvature, whereas inflation shocks matter most for the level of the yield curve. They also find that the forecast performance of an affine model with macro factors is better than that of a model with only latent factors. Piazzesi (2005) shows that the slope of the yield curve is driven by shocks to monetary policy.⁵ Outside the class of affine no-arbitrage models, Diebold and Li (2006) build on Nelson and Siegel (1987) to develop a method to estimate dynamic yield curve factors precisely for each period from yield data. Several recent papers apply this methodology to study the impact of macro factors on the yield curve and on bond

⁴Throughout the paper, we use the terms "term premium" and "bond risk premium" synonymously.

⁵There are, of course, other approaches relying on macro-finance linkages based on more standard equilibrium models which also serve to capture the failure of the Expectation Hypothesis and the existence of time-varying risk premia. For example, Wachter (2006) and Buraschi and Jiltsov (2007) study equilibrium models with habit-formation as in Campbell and Cochrane (1999). Bansal and Shaliastovich (2008) and Rudebusch and Swanson (2008) build models with long-run risks based on Bansal and Yaron (2004) to capture failures of the EH and the existence of time-varying bond risk premia. Buraschi and Jiltsov (2005) study risk premia in a money-augmented real business cycle model with taxes and endogenous monetary policy.

risk premia. Diebold, Rudebusch, and Aruoba (2006) examine the dynamic interaction between macro factors and the yield curve, finding strong evidence for effects of macro on yields but also for effects running from yields to macro variables. Diebold, Li, and Yue (2008) study global yield curve factors which are related to the global business cycle and have resemblance to worldwide inflation and real activity. We contribute to this active literature by directly investigating the relationship between these yield curve factors and the expectations of individual forecasters about future term premia movements.

Our approach of relying on survey information is not at all uncommon in the bond literature. Chun (2009) also uses analyst forecasts on GDP, inflation, and the Federal Funds rate to link fluctuations in bond yields to expectations about monetary policy and macro conditions. Thus, he studies the impact of forward-looking macro expectations on bond yields, an approach we also follow in this paper.⁶ Piazzesi and Schneider (2009) use median forecasts along with predictive regressions to disentangle (aggregate) subjective risk premia and prediction errors of professional forecasters. Wright (2009) uses survey information on long-term inflation, GDP, and interest rates to construct term premium estimates. He also studies the effect of inflation and output uncertainty on risk premia in a panel of countries. Wright ascribes a large amount of the variation in term premia to role of inflation uncertainty. This result seems to make sense, since Piazzesi and Schneider (2006), Campbell, Sunderam, and Viceira (2009), or Rudebusch and Swanson (2008) all argue that inflation uncertainty matters for term premia, a point we pay special attention to in this paper. Söderlind (2009) uses survey information to construct proxies for inflation and output growth uncertainty and finds that uncertainty (as well as liquidity factors) is a significant driver of bond risk premia over the period from 1997 to 2008. Söderlind also finds that output growth uncertainty lowers the term premium whereas inflation uncertainty increases it.

While our study is related to all these papers, we go beyond the existing contributions in the following way: We propose a proxy for term premium expectations which is basically model-free, real-time and easily implementable, and explore a panel of individual forecasters to relate these

⁶However, Chun does not investigate individual forecasters or the impact of inflation and output growth uncertainty.

proxies to macroeconomic *expectations*, measures of aggregate macro uncertainty, and measures of real-time macro activity. To the best of our knowledge, we are the first to analyze these issues in a comprehensive and coherent approach.

3 Measuring Term Premium Expectations

There are a number of ways to derive term premia from surveys which differ with respect to their data requirements and/or their reliance on expected bond returns versus bond yields (see e.g. Piazzesi and Schneider, 2006, 2009; Wright, 2009, for different approaches). We propose a simple way to calculate term premium expectations of individual forecasters that can be readily implemented with minimal data requirements and mild assumptions and approximations. The construction of our term premium expectation proxy is described next.

Construction of our proxy. The Expectation Hypothesis (EH) implies that long-term yields on zero-coupon bonds are equal to the average of the expected future short-term interest rates and a constant term premium (Campbell and Shiller, 1991),

$$y_t^n = \pi + \frac{1}{k} \sum_{i=0}^{k-1} E_t[y_{t+mi}^m], \tag{1}$$

where y_t denotes a log yield measured in quarter t, n (m) denotes the quarters to maturity of the long-term bond (of a T-bill), k = n/m, and π equals the non-varying term premium of holding the long-term bond. Since numerous studies have documented that the EH does not hold due to timevarying term (or risk) premia (e.g. Fama and Bliss, 1987; Campbell and Shiller, 1991), we follow Cochrane and Piazzesi (2008), introducing a time subscript t to π in Equation (1). Considering T-bills with a time to maturity of exactly m = 1 quarter as the short rate, we formulate the relationship between long-term and short-term interest rates as

$$y_t^n = \pi_t + \frac{1}{n} \sum_{i=0}^{n-1} E_t[y_{t+i}^1], \qquad (2)$$

and in terms of expectations for the interest rate h quarters ahead, we have

$$E_t[y_{t+h}^n] = E_t[\pi_{t+h}] + \frac{1}{n} \sum_{i=0}^{n-1} E_t[E_{t+h}[y_{t+h+i}^1]].$$
(3)

Using the law of iterated expectations and taking differences of Eqs. (3) and (2), we obtain the expected changes in long-term yields

$$E_t[y_{t+h}^n] - y_t^n = E_t[\pi_{t+h}] - \pi_t + \frac{1}{n} \sum_{i=0}^{n-1} E_t[y_{t+h+i}^1] - \frac{1}{n} \sum_{i=0}^{n-1} E_t[y_{t+i}^1].$$
(4)

Rearranging terms yields the expected change in term premia $E_t[\Delta \pi_{t+h}]$, which is given by

$$E_t[\Delta \pi_{t+h}] = E_t[\pi_{t+h}] - \pi_t = E_t[y_{t+h}^n] - y_t^n - \left(\frac{1}{n}\sum_{i=0}^{n-1} E_t[y_{t+h+i}^1] - \frac{1}{n}\sum_{i=0}^{n-1} E_t[y_{t+i}^1]\right).$$
(5)

As there are overlapping time periods in the two sum operators (in parentheses on the RHS), some of the expected future short rates cancel out. Choosing h = 1 for simplicity, Eq. (5) can be written as

$$E_t[\Delta \pi_{t+1}] = E_t[\pi_{t+1}] - \pi_t = E_t[y_{t+1}^n] - y_t^n - \frac{1}{n} \left(E_t[y_{t+n}^1] - y_t^1 \right), \tag{6}$$

which is the expected change in long-term yields minus the difference of the "corner" short-term interest rates (the one which is expected today for the period following the maturity of the long-term bond $(E_t[y_{t+n+1}^1])$ as well as the current one (y_t^1)).

Note that the number of overlapping time periods which cancel out in the two sum operators decreases when a larger forecast horizon h is considered. As a consequence, the expected change in the risk premium includes the difference between the sum of the h "corner" interest rates on each side. For example, for a horizon of h = 2, (5) reads

$$E_t[\Delta \pi_{t+2}] = E_t[y_{t+2}^n] - y_t^n - \frac{1}{n} \left(E_t \left[y_{t+n+1}^1 + y_{t+n}^1 \right] - E_t[y_{t+1}^1] - y_t^1 \right)$$
(7)

and similarly for longer horizons. Hence, our measure relates to expectations about future changes

in term premia and is just the expected yield change of a long-term bond plus a minor adjustment for (expected and current) short rates.

Empirical construction of the proxy. Based on Eqs. (6) and (7), we calculate the expected change in term premia from forecasters' expectations about long-maturity and short-maturity yields. We consider T-bond yields (10 years to maturity) from the SPF to obtain the expectation value in the first component on the RHS in Eqs. (6) and (7), $E_t[y_{t+h}^n] - y_t^n$. Of course, yield expectations in the SPF do not apply to zero-coupon bonds, but we stress that we are examining expectations about *yield changes* and not the level of yields. The yield change of a zero-coupon bond is likely to be much better approximated by the yield change of a coupon bond compared to approximating yield levels of zero-coupon bonds by coupon bonds. In fact, the correlation of zero-coupon bond yield changes in the Gürkaynak, Sack, and Wright (2007) data and changes of T-bond yields (from the FED St. Louis) is approximately 93%. Thus, working with expected yield changes for T-bonds from the SPF does not seem to be a terrible approximation.

To complete the computation of $E_t[\Delta \pi_{t+h}]$ in Eqs. (6) and (7), we further need to identify the expected short-term interest rates in the second component on the RHS. The expected short rates for subsequent quarters $(E_t[y_{t+h}^1])$ are directly available in the SPF data. However, the dataset does not contain subjective information about the expected short-term interest rates in the distant future $E_t[y_{t+n}^1]$. Therefore, we assume that forecasters expect short-term interest rates in the distant future to equal the unconditional mean of the short-term interest rates for the time period 1981 to the current point in time t, which implies that forecasters rely on past long-run averages when it comes to long-run forecasting.⁷ We find this a reasonable assumption.⁸ Furthermore, we also note that the expression with the short-term interest rate differences is multiplied by 1/n

 $^{^{7}}$ For robustness, we also consider the unconditional mean of the short-term interest rates for the time period 1981 to 2009 and do not find important changes.

⁸In our setup, the long-term bond has a maturity of 10 years. Therefore, it is plausible that forecasters have only vague ideas about short-term interest rates 40 quarters ahead, and thus basically forecast the recursive unconditional mean. In this vein, as long as the short-rate is stationary, standard time-series models will also deliver a forecast close to the unconditional mean when iterated 40 periods into the future. In the same vein, standard affine term structure models with no-arbitrage restrictions generally also have an inherent tendency to produce forecasts equal to the recursive unconditional mean (see also the discussion in Cochrane and Piazzesi, 2008).

(which equals 1/40 in our case, since we are working on a quarterly frequency and consider ten year maturities). Hence, the expected change in bond yields dominates the expression on the RHS of Equations (6) and (7), such that we do not expect the results to be driven by the identifying assumption about long-run forecasts for the short rate.

In short, we are relying on a combination of expected changes in yields of long and short maturities to obtain an observable proxy for expected changes in term premia. While we have to make some simplifying assumptions, these do not appear to be overly strong and they certainly do not drive our results. Advantages of our proxy are that it can be easily computed in real-time and has no hindsight bias, that it can be constructed for average survey expectations or individual forecaster expectations, and that it is directly observable and does not have to be estimated.

Interpretation of the proxy. Now, what does "expected change in term premia" mean in economic terms? Our term premium expectation factor captures information about *future changes* in term premia, so that our results below can not be interpreted in the same way as in many other papers where macro factors (or other proxies for business cycle risk) ought to capture the *current levels* of risk premia. For instance, the point in Ludvigson and Ng (2009) is to find business cycle state variables which measure contemporaneous levels of risk premia. Hence, a higher risk premium today signals high required returns and should thus translate into high returns going forward. In our case, we investigate a proxy for future changes in risk premia which is a different concept: expectations about positive risk premium changes imply that required returns, i.e. discount rates, will rise in the future (without making statements about current levels of risk premia) and should thus translate into lower returns in future periods (as prices fall due to increases in risk premia). Hence, our results are not directly comparable with those of earlier papers investigating drivers of contemporaneous risk premia and we will comment on these different interpretations below when presenting our empirical results.

4 Data and Empirical Approach

4.1 Data Sources and Variable Construction

Data description. Our analysis of the determinants of individual investors' term premium expectations requires expectations about future bond yields and macro variables. We rely on the *Survey of Professional Forecasters* (SPF) to obtain these micro-level expectations. The SPF covers participants from financial firms, banks, consulting firms, or research centers. The average participation is about 38 forecasters per quarter. We choose the SPF because it contains our variables of interest, because it allows us to calculate sensible measures of forecaster uncertainty (see below), and because its use as a data source is widely established in academic studies (e.g. Ang, Bekaert, and Wei, 2007). Our sample covers 70 quarters from 1992Q1 to 2009Q2 and a total of 153 different forecasters.⁹ Of course, timing issues arise when using survey data, and we detail the timing and our approach of aligning survey expectations with other data in the Appendix (see Appendix A.1 and A.2).

We obtain the expected change in the term premium for each forecaster from his or her predictions of 10-year Treasury Bond Yields and 3-months T-bill rates for the subsequent quarters, see Eq. (6) or (7) above. We also include the expectations about real GDP and inflation as explanatory variables in our analysis. For real GDP, we calculate the expected (log) growth rate, i.e. the forecast relative to the nowcast for the current quarter, so that we look at expected output growth. The expected inflation rate is included in levels.

Uncertainty measures. We also derive measures of uncertainty from the density forecasts about real GDP (PRGDP) and inflation (PRPGDP) of the survey. These survey questions ask the forecasters to indicate what probabilities they ascribe to each of ten possible ranges of percentage changes of the GDP levels as well as the price level in the current year and the next year,

⁹The SPF did not include questions about Treasury yield expectations before 1992.

respectively. The lower and upper category are open-ended. We compute empirical moments of the individual distributions as follows: we consider the outer ranges as closed categories with a midpoint which is equally spaced to the other midpoints in the scale. Based on the midpoints of all categories, we compute the individual means $\theta_{t,i}$ and variances $\sigma_{t,i}^2$ for the probability distributions at each point in time.¹⁰ We adjust the cross-sectional average variance $\sigma_t^2 = \frac{1}{N} \sum_{i=1}^{N} \sigma_{t,i}^2$ for seasonality by substracting a season-specific average of σ_t^2 from the original values. The adjusted series serve as time-varying measures of aggregate uncertainty about GDP ($\Psi(RGDP_{TY}), \Psi(RGDP_{NY})$) and the inflation rate ($\Psi(INF_{TY}), \Psi(INF_{NY})$) in the current (TY) and the next year (NY), respectively.

Figure 1, Panel (a), shows a time-series plot of our proxies for inflation and GDP uncertainty. Note that our procedure of measuring uncertainty yields somewhat different results than earlier papers. For example, Wright (2009) finds that uncertainty (measured as forecast dispersion) increases heavily during the financial crisis in 2007 to 2009. We also find increased uncertainty before the outbreak and at the beginning of the crisis, but uncertainty quickly decreases as forecasters become quite certain that GDP growth and inflation rates will fall. Thus, part of the large forecast dispersion in Wright (2009) seems to stem from forecasters' disagreement, but not necessarily from their uncertainty.

Panel (b) of Figure 1 shows a scatter plot of inflation and output growth uncertainty. As may be expected, the two series are positively correlated (e.g., $Corr(\Psi(\text{RGDP}_{NY}), \Psi(\text{INF}_{NY})) = 0.5)$. Thus, we compute an orthogonalized measure of inflation uncertainty, namely the residuals of the regression of $\Psi(\text{INF}_{NY})$ on $\Psi(\text{RGDP}_{NY})$ and a constant. The resulting time series (denoted $\Psi(\text{INF}_{NY})^{\perp}$) has a high correlation of 0.86 with the unadjusted inflation uncertainty measure $\Psi(\text{INF}_{NY})$ but is uncorrelated with real GDP uncertainty by construction.

FIGURE 1 ABOUT HERE

¹⁰Our choice of inferring uncertainty from density forecasts and not from cross-sectional point forecast dispersion (as e.g. in Wright, 2009) is based on findings in the literature which suggest that forecast dispersion may be a somewhat crude proxy for uncertainty (although the two measures are correlated, of course). For the sake of brevity, we refer to Giordani and Söderlind (2003), Lahiri and Sheng (2010), Liu and Lahiri (2006), and Pesaran and Weale (2006) for details on these issues.

Real Time Data. For our empirical analysis, we also make use of real time yield data (mainly based on the data by Gürkaynak, Sack, and Wright (2007)), which are also used to construct yield curve factors as in Diebold and Li (2006) and the bond factor of Cochrane and Piazzesi (2005), denoted *CP*. Furthermore, we rely on real-time data on real GDP growth, CPI inflation, industrial production (IP) and money growth (in M2), as well as CRSP bond returns for several maturities. To save space, however, we detail the data sources, variable constructions and timing issues in the appendix to this paper (see Appendix A.1 and A.2).

4.2 Empirical Approach

We are interested in the determinants of expected term premium changes of individual investors. These determinants include other forward-looking variables (individual macro expectations) which have to be treated as being endogenous, as well as variables that can be considered exogenous. We thus specify our general (dynamic) panel regression model as

$$e_{i,t} = a_1 e_{i,t-1} + \Xi_{i,t} \gamma + \Psi_t \delta + Z_t \beta + \epsilon_{i,t} \tag{8}$$

where $e_{i,t}$ is a shortcut for the expected change in term premia as defined in Eqs. (6) and (7) (i.e., $e_{i,t} \equiv E_{i,t}[\Delta \pi_{t+h}])$.¹¹ $\Xi_{i,t} \equiv E_{i,t}[X_{i,t+h}]$ denotes a vector of subjective expectations of forecaster *i* about macroeconomic variables such as expected output growth and expected inflation, vector Ψ_t collects measures of aggregate uncertainty about future output growth and inflation rates, and Z_t denotes a vector of additional exogenous control variables. Our interest centers on the effect of expected macro movements $\Xi_{i,t}$ and uncertainty about macro movements Ψ_t on individual expectations about future bond risk premia $e_{i,t}$. Lagged (expected) risk premium changes and other observed macro factors in Z merely serve as control variables or are included to highlight additional aspects regarding the relation of our term premium expectations with other well-known

¹¹Throughout this paper, we report the results for the forecast horizon of h = 1 quarter (see the Web Appendix for the results for h = 4). For this horizon, $e_{i,t}$ is composed of an forecaster-specific component $E_{i,t}[y_{t+1}^n]$ minus a common component $\omega_t = y_t^n - \frac{1}{n} \left(E_t[y_{t+n}^1] - y_t^1 \right)$.

factors. The specification of the error term $\epsilon_{i,t} = \alpha_i + \mu_{i,t}$ takes into account that forecasters' expectations may exhibit unobserved heterogeneity. Hence, we work with a fixed effects setting and investigate time variation in term premia.

In Eq. (8), we regress current expectations about future risk premium changes $e_{i,t}$ on expectations about other macroeconomic variables $E_{i,t}[X_{i,t+h}]$ to single out the effect of expected macro movements on bond risk premia. While this approach is natural for our analysis, it generates a potential endogeneity problem since there is no reason to assume that causality strictly runs from $E_{i,t}[X_{i,t+h}]$ to $e_{i,t}$ and not vice versa. To tackle this challenge we rely on instrumental variable estimators for all our main results and instrument for current macro expectations with lagged macro expectations. We do this within the Generalized Method of Moments (GMM) framework of Arellano and Bond (1991). Furthermore, we take care of potential problems arising from the inclusion of too many instruments in panel regressions with a large time dimension relative to the cross-sectional dimension of the panel, and we account for autocorrelation and heteroskedasticity in our inference. Details on the exact estimation of the dynamic panel data model, the choice of instruments, and computation of standard errors are delegated to the appendix of this paper (see Appendix A.3). We have also checked that our results are not driven by the specific estimation method of Arellano and Bond (1991) and obtained findings very similar to those reported below in a, e.g. 2SLS panel or pooled regression framework. In this context, we have also used lagged realized macroeconomic time series (e.g., GDP and inflation) in some specifications to instrument for the lagged dependent variable and the endogenous variables to circumvent potential problems with measurement errors on both sides of Eq. (8).

5 Results

5.1 Properties of Term Premium Expectations

To set the stage, we plot time-series of aggregated expected term premium changes for horizons of one to four quarters in Figure 2 (red, solid line). We also show the "*realized term premium changes*" (blue, dashed line), which are computed by simply replacing expected log yield in Eq. (6) with actual future log yields. As one may expect, it can be seen that *expected* term premia are quite persistent and seem to be less volatile than *realized* changes in term premia. Compared to ex-post realizations of bond returns (or yield changes), the real time expectations about term premium changes appear to be a less noisy measure and should thus serve as a useful proxy to study the link between macro factors and term premia.

As a final note, there is a large decline in term premium expectations towards the end of our sample, starting with the onset of the financial crisis in 2007. This result is well in line with a 'flight-to-quality' effect and is also found in Wright (2009), lending some credence to the relevance of our proxy for term premium expectations.

FIGURE 2 ABOUT HERE

Table 1 shows descriptive statistics for *expected* term premium changes on the left side and for *realized* term premium changes on the right side for comparison. Both are negative on average. This confirms earlier analyses showing a decline of term premia in advanced countries over the time period of our sample (Wright, 2009). Furthermore, the standard deviations shown in Table 1 validate the perception that *expected* term premium changes are less volatile than *actual* changes. This effect becomes more pronounced for longer forecast horizons h.

TABLE 1 ABOUT HERE

5.2 Macro Expectations and Aggregate Uncertainty

As noted above, our main interest lies in the impact of macro expectations and uncertainty of individual term premium expectations as specified in Eq. (8). Thus, we now proceed to estimate dynamic panel regressions with fixed effects via GMM. We regress forecaster-specific term premium expectations on lagged term premium expectations, forecaster-specific expectations about output growth and inflation, as well as aggregate uncertainty about output growth and inflation and report our results in Table 2 for various combinations of explanatory variables.

We robustly find that individual term premium expectations are positively autocorrelated: the lagged dependent variable has a coefficient of about 0.25-0.40 across specifications, which is highly significantly different from zero. We control for this persistence in all future regressions by including the lagged expected term premium change as a regressor.

Perhaps more interestingly, we find that expected real output growth $(E_{t,i}[\Delta \text{RGDP}])$ has a significantly positive impact as well, so that higher growth expectations induce forecasters to expect the term premium to rise. This result supports findings in Ludvigson and Ng (2009) that real macro activity is a strong time-series predictor of bond excess returns. The strong evidence in our study reinforces the view that real factors are an important driver of bond risk premia. It should be noted, however, that Ludvigson and Ng (2009) find that return forecasts are high when current real activity is low and interpret this as a countercyclical bond risk premium. Our findings suggest that low expected output growth makes forecasters expect lower term premium changes going forward. As explained earlier, this is not a contradiction to our findings: as declining risk premia in the future imply higher returns going forward, the two results are actually compatible in terms of their economic effects. The difference is thus one of interpretation and not of economic outcomes.¹²

We also find a positive coefficient for expected inflation $(E_{t,i}[INF])$. However, the impact of

¹²Also, this difference is not driven by using expectations (our study) instead of current output growth (as in Ludvigson and Ng). We show below that current output growth is positively related to expectations about changes in future term premia as well.

expected inflation on term premium changes is not significant in all specifications and becomes unimportant once we include uncertainty measures or include it jointly with GDP expectations. At first sight, this result seems surprising since inflation is considered to be a prime candidate for driving term premia. Earlier papers usually see a large role of inflation in determining bond yields (see e.g. Ang and Piazzesi, 2003; Diebold, Rudebusch, and Aruoba, 2006; Rudebusch and Wu, 2008). However, we are investigating expected *changes* in term premia whereas most earlier papers show that inflation relates to the *level* of bond yields and risk premia. Furthermore, one has to bear in mind that our sample period (starting in 1992) is not one of particularly high inflation rates. In this specific macroeconomic setting, it may well be that inflation levels are relatively less important than real growth or uncertainty.¹³

TABLE 2 ABOUT HERE

Turning to our uncertainty measures, we find that uncertainty unambiguously leads forecasters to raise their expectations about future term premia, i.e. higher aggregate uncertainty in the current quarter leads forecasters to expect capital losses on long-term bonds in the future. Interestingly, we find that both output growth uncertainty and inflation uncertainty are significant drivers of term premium expectations even when we include both uncertainty sources simultaneously (specification (ix)) by using the orthogonalized inflation uncertainty series. Strikingly, this finding indicates that both "*real uncertainty*" and "*nominal uncertainty*" (about real and nominal macro factors), by themselves, matter. Wright (2009) also finds that inflation uncertainty matters, but does not ascribe a large role to output uncertainty. Similarly, Söderlind (2009) finds that inflation uncertainty has a positive impact on term premia, but also that higher output uncertainty lowers risk premia. Our results – which are based on a different concept of measuring uncertainty as well as a different sample period – suggest otherwise.

Regarding the economic significance of our explanatory variables, we find (based on the joint specification (ix) in Table (2)) that the long-run impact after taking into account the autoregressive

¹³This result is, again, in line with Ludvigson and Ng (2009) who find that inflation is much less important than real activity.

effects is about 20 basis points for a one-standard deviation shock to expected real GDP growth, about 5 basis points for a one-standard deviations shock to expected inflation, and about 3 basis points for the two uncertainty measures. While these effects may appear small at first sight, one has to put this into the perspective of an unconditional standard deviation of "only" 40 basis points of the dependent variable, i.e. expected term premium changes. Thus, a rise of one standard deviation in expected real output growth has an effect that makes up for about 50% of the standard deviation of expected term premium changes and the other three determinants still have an impact of about 7%-15% relative to a typical movement in the dependent variable. Thus, expected real output growth has a rather large impact on expected term premium changes, whereas expected inflation and uncertainty are still economically significant but clearly less important.¹⁴

We also report Pseudo R^2 s and a couple of diagnostic statistics. We see that R^2 s are around 35% in specifications including uncertainty and/or output growth expectations and are somewhat lower if only inflation expectations are included. The *J*-test is far from rejecting the overidentifying restrictions and residuals seem largely free from autocorrelation (we test for second-order autocorrelation since we have a fixed-effects setting), except for the last three specifications which are significant at the 10%-level only.

5.3 Yield Curve Factors and CP-Factor

Given the prominence of yield curve factors in the literature, we next look at the relation of level, slope, and curvature (obtained as in Diebold and Li, 2006) with our proxy for term premium expectations.¹⁵ We also include the *CP Factor*, which has been proposed by Cochrane and Piazzesi (2005) to forecast excess returns of bonds of different maturities. Given that subjective term premium expectations have predictive power, they are also related to future bond returns. Hence, the *CP*-Factor should be able to explain expected changes in the term premium to some extent.

¹⁴As a robustness check, we provide results for specifications with a recession dummy (Table A.II) and for regressions where we use expectations for annual horizons, i.e. h = 4, in Table A.III in the Web Appendix to this paper. It can be seen from these robustness tests that our results are not driven by events of the recent financial crisis.

¹⁵Note that the Diebold and Li procedure results in a "slope" factor that has an almost perfectly negative correlation with the term spread. Thus, we multiply our slope factor with -1 so that a high slope means a steep yield curve.

A plot of the four factors is shown in Figure 3. The plot shows the decline of the level of yields during the 1990s and 2000s (Panel (a)), the inverted yield curves prior to the last two recessions (Panel (b)), and the curvature factor in Panel (c), which is similar to the average subjective term premium expectations in Figure 2. In fact, curvature and term premium expectations have a positive correlation of about 67% (see Table A.III in the Web Appendix). This result seems especially noteworthy in the light of findings in Cochrane and Piazzesi (2008), who report that curvature is linked to *future* expected returns (as opposed to current term premia). This is exactly what our expected term premium proxy ought to capture as well. Finally, the *CP*-Factor in Panel (d) seems to be rather unrelated to the three other yield factors, as already motivated in Cochrane and Piazzesi (2005).

FIGURE 3 ABOUT HERE

In order to learn about the relationship of these yield factors with forecasters' expectations, we again run dynamic panel regressions of individual term premium expectations on lagged individual expectations as well as the four factors and report our results in Table 3.¹⁶ Specifications (i) to (iv) show that each factor has a highly significant impact when included individually. The relation is such that a higher level, slope, or *CP*-Factor leads to lower risk premium expectations, whereas curvature has a positive impact on expected term premia. When including all three Diebold and Li (2006) factors, slope becomes insignificant but we obtain the highest Pseudo R^2 of 47%.

The estimated signs of factors seem to make sense when comparing them with earlier literature. For example, Diebold, Rudebusch, and Aruoba (2006) find that the *level* factor captures *inflation* (also see Rudebusch and Wu, 2008). To the extent that forecasters have mean-reverting expectations anchored at some level of inflation, one would expect a high level factor (i.e. high inflation) to be accompanied by high contemporaneous term premia but lower term premia expectations for the future.¹⁷ A similar argument may be made for the *slope*, for which Diebold, Rudebusch, and

¹⁶We include the lagged dependent variable to account for autocorrelation in the dependent variable only.

¹⁷We provide additional evidence on this later in the paper.

Aruoba (2006) argue that it may be interpreted as a proxy for *real activity*. While Diebold, Rudebusch, and Aruoba (2006) find that *curvature* seems rather unrelated to macro factors, our results and the discussion above suggest that curvature is highly correlated with expectations about future bond risk premia. We also see this positive relation in the panel regressions. Finally, the CP factor is negatively related to expectations about future term premia. This relation seems reasonable as Cochrane and Piazzesi (2005) show that CP is positively related to future bond excess returns, i.e. declining yields in the future. Note again that low term premium expectations and high return expectations for holding long-term bonds are essentially two sides of the same coin: if term premia are expected to decrease in the future, bond prices in the future will be expected to increase, i.e. investors will make capital gains when they sell long-term bonds prior to maturity. Cochrane and Piazzesi (2005), Ludvigson and Ng (2009) and others find strong evidence that return risk premia are countercyclical. In this paper, we show that this *countercyclical nature* of return risk premia corresponds to a *procyclical* behavior of term premia expectations: as shown above, high GDP growth expectations today are associated with term premia expected to be higher in the future. In the same vein, it also is reassuring to see the CP regression-based results reflected in the actual expectations of individual investors.

TABLE 3 ABOUT HERE

Since information in the term structure is related to the business cycle (see Estrella and Hardouvelis, 1991; Ang, Piazzesi, and Wei, 2005, among many others) and since the *CP* factor is informative about future macro conditions (Koijen, Lustig, and Van Nieuwerburgh, 2010), we also present results from joint specifications where we include macro expectations, uncertainty, and bond factors in Table 4. These results aim for a closer investigation of possible relationships between yield-related and macro factors.

Table 4 shows that level, slope, curvature, and the CP factor are significantly different from zero in all specifications. Interestingly, we find that level and slope both drive out aggregate uncertainty (whereas curvature and CP do not), suggesting that these two factors also capture uncertainty about real and nominal variables. This result seems novel to the literature and may shed some more light on the economic forces underlying these popular yield curve factors. It is also worth noting that expected real GDP growth is not driven out by any combination of other factors and even remains significant in the full specification (v). This underlines the role of expected output growth in the expectation formation with respect to movements in term premia.

TABLE 4 ABOUT HERE

5.4 Impact of Risk-Related Variables

We next turn to the question whether variables related to *risk* drive term premium expectations. In this context, we consider measures of realized yield volatility (for both TBOND yields and TBILL yields) and the BAA-AAA risk spread (based on Moody's yield data) as proxies. We report the results in Table 5. We find that all three measures of risk have a negative impact on expected term premium changes.¹⁸

TABLE 5 About here

The negative relation between risk and term premium expectations points at the existence of *flight-to-quality effects*: higher risk leads to an expected portfolio rebalancing into safe government securities which drives up prices and lowers term premia on these bonds in the near future. In fact, since our sample covers the major financial crisis of 2007 - 2009, this finding seems reasonable. Again, our results for macro expectations and aggregate uncertainty are more or less unaffected by controlling for these risk factors. We still find a strong and positive impact of expected output growth on term premia and a positive impact of uncertainty, although the latter varies somewhat across regression specifications.

We also find it noteworthy that risk-related variables and uncertainty have opposite effects on expected term premium changes. This suggests that caution should be exercised in empirical

¹⁸We have experimented with other risk proxies, such as recession dummies (also see Table A.II) and found very similar results.

applications where the two concepts are often treated as being close substitutes (see Anderson, Ghysels, and Juergens, 2009, for a somewhat related finding on the difference between risk and uncertainty in equity markets).

5.5 Real-time Macro Factors and Expected Term Premium Changes

As a final test, we also include real-time macro factors in our regressions to find out whether focusing on *expected* macro conditions yields additional insights relative to relying on *current* macro conditions. We rely on real-time vintage data and include growth rates of industrial production, GDP growth, CPI inflation, and real money growth (M2). Results for various specifications are collected in Table 6.

Our estimates show that current real-time CPI inflation has a significantly *negative* impact on term premium expectations at the micro-level, whereas expected inflation has a positive impact. This result is rather counterintuitive but may be understood by forecasters' reliance on meanreversion and anchored inflation expectations. To the extent that relatively high inflation leads forecasters to expect lower inflation in the future, this negative coefficient is in line with our finding that high expected inflation increases term premium expectations as shown above. This finding, however, underlines the importance of investigating forward-looking drivers of term premia rather than determinants mirroring the current state of the economy, as these two approaches may yield very different results.

Having said this, we find that output growth (growth of industrial production and GDP) has a significantly positive impact on expected term premia, just as our measure of expected output growth does. Thus, relying on expected as well as actual macro conditions leads to similar results. However, our estimates also reveal that actual, real-time output growth does not drive out expected output growth and the latter is still highly significant in all specifications examined in Table 6. Once again, these results emphasize that there is a role for forward-looking macro factors when modelling term premia over and above the information contained in the current state of the economy.

TABLE 6 ABOUT HERE

5.6 Predictive Regressions

Our results in the previous sections deal with individual expectations about future term premium movements and the impact of macro expectations and uncertainty on these risk premium expectations. While we believe that this approach provides valuable insights into how macro factors affect term premia, it is also of interest to see whether our term premium expectations are actually related to future bond returns or whether forecasters' expectations are merely an unimportant sideshow for bond markets. A link between future bond excess returns and expectations of forecasters with respect to term premium movements would clearly strengthen the case for our findings. To shed some light on this issue, we run predictive regressions as in Cochrane and Piazzesi (2005) or Ludvigson and Ng (2009)

$$rx(m)_{t+h} = \alpha_h + \beta_h \overline{e}_t + \varepsilon_{t+h} \tag{9}$$

of future bond excess returns rx(m) on current average term premium expectations \overline{e}_t , i.e. we use the average forecast across individuals discussed in Table 1 and Figure 1 above. The forecast horizon h varies from one to four quarters and we match the forecast horizon of returns with the forecast horizon of subjective term premium expectations. m denotes the maturity of the bonds underlying the excess returns and we also include average excess returns across all maturities (denoted rx(avg)). We report t-statistics based on Newey and West (1987) standard errors (with h lags for robustness) and also based on Hodrick (1992) standard errors which were shown to have better size properties when forecasting with persistent regressors (see Ang and Bekaert, 2007).¹⁹

If term premium expectations matter for bond markets, we would expect to see a significant impact of expectations on future bond returns, i.e. that the coefficient β_h is significant in Eq.

¹⁹Average term premium expectations have an autocorrelation of "only" 70%, see Table 1, which is not very high relative to other predictors such as dividend yields in stock markets. Thus, we expect finite-sample biases (Stambaugh, 1999) to be fairly mild in our sample.

(9). More specifically, we would expect to find a negative coefficient since higher term premia in the future imply that future bond yields must rise (relative to the short-rate) so that actual bond excess returns have to be lower in the future.

Results from these predictive regressions for maturities of m = 1, 2, ..., 5, 10 years and the average excess return over all maturities rx(avg) are provided in Table 7. Note that we only include data until 2007Q2 since there are huge return movements due to the subsequent financial crisis and these outliers drive much of our result given that we have a relatively short time-series.²⁰ We thus limit our analysis to "normal" situations and exclude extreme events.

Given this caveat, our results show that term premium expectations forecast bond excess returns with rather low R^2 s of up to 4% at a quarterly forecast horizon, but with R^2 s of up to 22% at an annual forecast horizon. The R^2 s as well as the levels of statistical significance (even when based on Hodrick (1992) standard errors) tend to increase with longer forecast horizons of up to four quarters. Predictive R^2 s are largest for excess returns over maturities between one to three years but we also find significant predictability at longer maturities. At an annual forecast horizon, expected term premium changes forecast average bond excess returns across maturities with an R^2 of around 13% and predictability is stronger for short horizons with R^2 s as high as 22% for one-year maturities. While other papers (Cochrane and Piazzesi, 2005) find much higher R^2 s (of up to 45%) for annual horizons (Cochrane and Piazzesi, 2005; Ludvigson and Ng, 2009) in longer samples, it should be noted that our factor is available in real-time and not estimated as in other papers. We will provide more details on this issue below.

TABLE 7 ABOUT HERE

Furthermore, we find a consistently negative predictive coefficient across forecast horizons and maturities as expected. Hence, our results suggest that forecasters' expectations contain relevant information for future bond returns and that movements in expected discount rates matter for

²⁰More precisely, our sample is short relative to sample sizes usually employed for predictive regressions. This fact is important, since actual returns are noisy and it is thus difficult to estimate expected returns from actual returns with sufficient precision. Thus, we choose not to include the extremely unusual market movements of the financial crisis in our regressions since these data points would swamp our results.

bond excess returns, especially at the shorter end of the maturity spectrum.

The question remains whether our proxy for term premium expectations captures the same information as other bond predictors known from the literature. As a comparison, we re-estimate our predictive regressions with a real-time (i.e. recursively estimated) CP-factor instead of our term premium expectations $E_t[\Delta \pi_{t+1}]$. Results are reported in the Web Appendix to conserve space (see Table A.V). We find that the CP factor works best for longer maturities and short forecast horizons (where it clearly dominates our factor) during our sample period whereas our factor performs best for longer forecast horizons and shorter maturities (where it outperforms the CPfactor). Hence, our proxy for term premium expectations seems to capture different information than the CP factor.²¹ The results we have discussed in the previous sections are therefore not only relevant to describe patterns of the (potentially irrational and subjective) expectation formation of professional forecasters, but may improve our understanding of risk premia movements in bond markets more generally.

6 Conclusions

We analyze *individual* expectations about term premium movements in a panel of forecasters and relate these individual expectations to expected real and nominal macro variables, to aggregate uncertainty about real and nominal macro variables, as well as to further control variables, such as term structure factors, risk-related factors, and real-time macro developments. A novel aspect of our analysis is our focus on the impact of inherently forward-looking macro factors on expectations about term-premia in a panel approach which allows for heterogeneity across forecasters. We find that individual forecasters' macro expectations are strongly related to expectations about bond risk premia, and we find the largest impact for real output growth and uncertainty about real output growth. Thus, there is a strong link between macro developments and term premia in bond

 $^{^{21}}$ We also find that our factor is different when we compare it to real-time measures of growth in industrial production. Ludvigson and Ng (2009) show that their real macro factor is related to industrial production growth so that this comparison again suggests that expected term premium expectations are different.

markets.

Furthermore, our results suggest that curvature of the term structure is strongly related to subjective expectations about term premia, while the level and slope factors seem to capture information similar to that contained in our uncertainty measures about future real output growth and inflation. We also show that focusing on expected *future* macro conditions can lead to different results than analyzing the impact of *current* macro conditions on risk premia, and that expected macro conditions contain information for term premia over and above the information contained in current (real-time) macro conditions. Finally, an aggregate measure of term premium expectations forecasts future bond returns over horizons of up to one year in a way that is consistent with the idea that our proxy for expected term premium changes forecasts future changes in discount rates.

References

- ANATOLYEV, S., AND N. GOSPODINOV (2010): "Specification Testing in Models with Many Instruments," *Econometric Theory*, forthcoming.
- ANDERSON, E. W., E. GHYSELS, AND J. L. JUERGENS (2009): "The Impact of Risk and Unvertainty on Expected Returns," *Journal of Financial Economics*, 94, 233–263.
- ANG, A., AND G. BEKAERT (2007): "Stock Return Predictability: Is it there?," *Review of Finan*cial Studies, 20(3), 651–707.
- ANG, A., G. BEKAERT, AND M. WEI (2007): "Do Macro Variables, Asset Markets, Or Surveys Forecast Inflation Better?," *Journal of Monetary Economics*, 54, 1163–1212.
- ANG, A., AND M. PIAZZESI (2003): "A No-Arbitrage Vector Autoregression of Term Structure Dynamics with Macroeconomic and Latent Variables," *Journal of Monetary Economics*, 50(4), 745–787.
- ANG, A., M. PIAZZESI, AND M. WEI (2005): "What Does the Yield Curve Tell Us About GDP Growth," *Journal of Econometrics*.
- ARELLANO, M., AND S. BOND (1991): "Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations," The Review of Economic Studies, 58(2), 277–297.
- BANSAL, R., AND I. SHALIASTOVICH (2008): "A Long-Run Risks Explanation of Predictability Puzzles in Bond and Currency Markets," *Working Paper*.
- BANSAL, R., AND A. YARON (2004): "Risk for the Long Run: A Potential Resolution of Asset Pricing Puzzles," *Journal of Finance*, 59(4), 1481–1509.
- BIKBOV, R., AND M. CHERNOV (2008): "No-arbitrage Macroeconomic Determinants of the Yield Curve," Working Paper, London Business School.
- BLUNDELL, R., AND S. BOND (1998): "Initial Conditions and Moment Restrictions in Dynamic Panel Data Models," *Journal of Econometrics*, 87(1), 115–143.
- BURASCHI, A., AND A. JILTSOV (2005): "Inflation Risk Premia and the Expectation Hypothesis," Journal of Financial Economics, 75, 429–490.
 - (2007): "Habit Formation and Macroeconomic Models of the Term Structure of Interest Rates," *Journal of Finance*, 62, 3009–3063.
- CAMERON, A. C., AND P. K. TRIVEDI (2005): *Microeconometrics: Method and Applications*. Cambridge: Cambridge University Press.
- CAMPBELL, J. Y., AND J. H. COCHRANE (1999): "By Force of Habit: A Consumption Based Explanation of Aggregate Stock Market Behavior," *Journal of Political Economy*, 107(2), 205–251.
- CAMPBELL, J. Y., AND R. J. SHILLER (1991): "Yield Spreads and Interest Rate Movements: A Bird's Eye View," *Review of Economic Studies*, 58, 495–514.
- CAMPBELL, J. Y., A. SUNDERAM, AND L. M. VICEIRA (2009): "Inflation Bets or Deflation Hedges? The Changing Risks of Nominal Bonds," *Working Paper, Harvard University*.
- CHUN, A. L. (2009): "Expectations, Bond Yields and Monetary Policy," Working Paper, Copenhagen Business School.
- COCHRANE, J. H. (2005): Asset Pricing (Revised Edition). NJ: Princeton University Press.
- COCHRANE, J. H., AND M. PIAZZESI (2005): "Bond Risk Premia," American Economic Review, 95(1), 138–160.
 - (2008): "Decomposing the Yield Curve," Working Paper, Chicago Booth.

- DIEBOLD, F. X., AND C. LI (2006): "Forecasting the Term Structure of Government Bond Yields," Journal of Econometrics, 130(2), 337–364.
- DIEBOLD, F. X., C. LI, AND V. Z. YUE (2008): "Global Yield Curve Dynamics and Interactions: A Dynamic Nelson-Siegel Approach," *Journal of Econometrics*, 146, 351–363.
- DIEBOLD, F. X., M. PIAZZESI, AND G. D. RUDEBUSCH (2005): "Modeling Bond Yields in Finance and Macroeconomics," *American Economic Review, Papers & Proceedings*, 95, 415–420.
- DIEBOLD, F. X., G. D. RUDEBUSCH, AND S. B. ARUOBA (2006): "The Macroeconomy and the Yield Curve: A Dynamic Latent Factor Approach," *Journal of Econometrics*, 131(1-2), 309–338.
- ESTRELLA, A., AND G. HARDOUVELIS (1991): "The Term Spread as a Predictor of Real Economic Activity," *Journal of Finance*, 46, 555–576.
- FAMA, E. F., AND R. R. BLISS (1987): "The Information in Long-Maturity Forward Rates," American Economic Review, 77, 680–692.
- GIORDANI, P., AND P. SÖDERLIND (2003): "Inflation Forecast Uncertainty," *European Economic Review*, 47, 1037–1059.
- GÜRKAYNAK, R. S., AND J. H. WRIGHT (2010): "Macroeconomics and the Term Structure," Working Paper.
- GÜRKAYNAK, R., B. SACK, AND J. WRIGHT (2007): "The US Treasury Yield Curve: 1961 to the Present," *Journal of Monetary Economics*, 54(8), 2291–2304.
- HANSEN, L. P. (1982): "Large Sample Properties of Generalized Method of Moments Estimators," *Econometrica*, 50(4), 1029–1054.
- HODRICK, R. J. (1992): "Dividend Yields and Expected Stock Returns: Alternative Procedures for Inference and Measurement," *Review of Financial Studies*, 5(3), 357–386.
- KIM, D. H. (2009): "Challenges in Macro-Finance Modeling," Federal Reserve Bank of St. Louis Review, 91, 519–544.
- KOIJEN, R. S. J., H. LUSTIG, AND S. VAN NIEUWERBURGH (2010): "The Cross-Section and Time-Series of Stock and Bond Returns," AFA 2010 Meetings paper.
- LAHIRI, K., AND X. SHENG (2010): "Measuring Forecast Uncertainty by Disagreement: The Missing Link," Journal of Applied Econometrics, 25, 514–538.
- LIU, F., AND K. LAHIRI (2006): "Modelling Multi-period Inflation Uncertainty Using A Panel Of Density Forecasts," *Journal of Applied Econometrics*, 21, 1199–1219.
- LUDVIGSON, S. C., AND S. NG (2009): "Macro Factors in Bond Risk Premia," *Review of Financial Studies*, 22(12), 5027–5067.
- NELSON, C. R., AND A. F. SIEGEL (1987): "Parsimonious Modeling of Yield Curves," Journal of Business, 60(4), 473–489.
- NEWEY, W. K., AND K. D. WEST (1987): "A Simple, Positive Semi-Definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix," *Econometrica*, 55(3), 703–708.
- PESARAN, M. H., AND M. WEALE (2006): "Survey Expectations," in *Handbook of Economic Forecasting*, ed. by G. Elliott, C. W. Granger, and A. Timmermann. North-Holland, vol. 1, chap. 14, pp. 715–776.
- PIAZZESI, M. (2005): "Bond Yields and the Federal Reserve," Journal of Political Economy Volume, 113, 311–344.
- PIAZZESI, M., AND M. SCHNEIDER (2006): "Equilibrium Yield Curves," *NBER Macro Annual*, 21, 389–442.
 - (2009): "Trend and Cycle in Bond Risk Premia," Working Paper, Stanford University.

- ROODMAN, D. (2009): "A Note on the Theme of Too Many Instruments," Oxford Bulletin of Economics and Statistics, 71, 135–158.
- RUDEBUSCH, G. D. (2010): "Macro-Finance Models of Interest Rates and the Economy," Working Paper, Fed San Francisco.
- RUDEBUSCH, G. D., AND E. T. SWANSON (2008): "Examining the Bond Premium Puzzle with a DSGE Model," *Journal of Monetary Economics*, 55, S111–S126.
- RUDEBUSCH, G. D., AND T. WU (2008): "A Macro-Finance Model of the Term Structure, Monetary Policy, and the Economy," *Economic Journal*, 118, 906–926.
- SÖDERLIND, P. (2009): "Inflation Risk Premia And Survey Evidence On Macroeconomic Uncertainty," Working Paper, St. Gallen.
- STAMBAUGH, R. F. (1999): "Predictive Regressions," Journal of Financial Economics, 54, 375–421.
- WACHTER, J. A. (2006): "A Consumption-based Model of the Term Structure of Interest Rates," Journal of Financial Economics, 79, 365–399.
- WINDMEIJER, F. (2005): "A Finite Sample Correction for the Variance of Linear Efficient Two-Step GMM Estimators," *Journal of Econometrics*, 126(1), 25–51.
- WRIGHT, J. H. (2009): "Term Premia and Inflation Uncertainty: Empirical Evidence from an International Panel Dataset," *American Economic Review*, forthcoming.

Appendix

A.1 Data

Timing of survey expectations. The SPF questionnaires are sent to the participants at the end of the first month of each quarter. As the deadline for returning the questionnaires is the middle of the second month of the respective quarter, the professional forecasters respond within a two-week time-frame. Based on this response procedure, we refer to the period from the last survey deadline to the current one as a *survey quarter*. Note that unlike the *target quarter*, which corresponds to the conventional calendar quarters, *survey quarters* are spaced from Nov/16-Feb/15, Feb/16-May/15, May/16-Aug/15 and Aug/16 to Nov/15 in each year.

Interest rates and yields. We collect three months US T-bill rates and US Corporate Bond Yields (by the Federal Reserve) from *Datastream*. Bond yields for longer maturities are taken from the smoothed US Treasury yield curve data provided by Gürkaynak, Sack, and Wright (2007). These daily time series can be transformed into several variables required by our analysis: We consider the *last* realization in January, April, July and October of T-bill rates and ten-year-tomaturity bond yields in the respective information sets (including for the computation of risk premia as indicated in Equations (6) and (7). In contrast, we take the *mean* of the daily short rates and 10 years bond yields to construct series of *realized* changes in term premia. We also compute the log change of interest rates between the first and the last day in a survey quarter ($\Delta TBOND$, $\Delta TBILL$) as well as the standard deviation of daily interest rates within a survey quarter ($\sigma(TBOND)$, $\sigma(TBILL)$), respectively.

Macroeconomic Variables. To operate with macroeconomic figures which have actually been available to the forecasters, we exploit the real time data collected by the *Federal Reserve Bank* of *Philadelphia*. In particular, we consider the US total industrial production index (IP), the consumer price index (CPI), the real gross domestic product (RGDP) as well as the nominal money stock (M2). The real time data set ignores future data revisions or redefinitions and facilitates a relatively accurate timing of the information inflow. As the SPF does not include the exact individual response date, we assume that economic figures are included in the forecasters' information set if they are released by the end of the first month in a quarter (January, April, July, October). We transform all variables to (log) year-over-year growth rates (except for inflation). As the CPI has only been available in the real time data since Q1 1994, we compute inflation rates from ex-post data for the previous period. We compute the yoy (log) real money growth by substracting the (log) yoy inflation rate from the (log) nominal money growth. The IP, CPI, and M2 are published by the releasing institutions on a monthly basis in the middle of the subsequent month. As a consequence, we consider the values for December, March, June, and September, respectively. As only the releases in February, May, August and November for the CPI and the money stock are available in the real time data set of the Federal Reserve Bank of Philadelphia, we rely on revised data for the first month of a quarter for these two variables. GDP is released in the second month of a quarter for the preceding quarter. Accordingly, the information set includes the GDP figure of the third quarter at the end of January, of the fourth quarter at the end of April, of the first quarter at the end of July, and of the second quarter at the end of October.

A.2 Construction of Bond Factors and Excess Returns

Level, slope, and curvature. Diebold and Li (2006) demonstrate that the three time varying parameters of an exponential components framework are suitable to represent the yield curve factors "level", "slope", and "curvature". This method allows us to estimate precise yield factors for each period without making use of data beyond the forecasters' information set. We compute these factors based on a monthly series of bond yields for different maturities (last trading days in the months). Note that unlike Diebold and Li (2006), who model unsmoothed Fama-Bliss bond yields, we estimate the loading factors for the maturities of 1,2,3...10 years based on the smoothed data from Gürkaynak, Sack, and Wright (2007). A comparison of our measures for level, slope, curvature, and those of Diebold and Li (2006) for the period 1971/01 to 2000/12 yields a correlation coefficient of 0.99 (level), 0.99 (slope) and 0.73 (curvature). Note that the "slope factor" from this procedure has been shown to be almost perfectly negatively correlated with an empirical slope factor (defined as the ten-year yield minus the three-month yield), such that it is high when short rates exceed long rates. To make our results more easily interpretable on conventional grounds, we multiply the slope factor with minus one so that high values indicate a steep yield curve and vice versa.

Cochrane-Piazzesi factor. We compute a monthly series of the excess return forecasting factors put forth by Cochrane and Piazzesi (2005) (the "CP-Factors") by a recursive strategy as follows: First, we transform the monthly series of bond yields into prices, forward rates and excess returns. To avoid multicollinearity in our regression of average excess returns on forward rates, we only keep the one, three and five-year forward rates on the RHS (following an approach proposed in Cochrane and Piazzesi (2008) to work with the smoothed bond yields in the data of Gürkaynak, Sack, and Wright (2007)). We ensure a real-time computation (avoiding potential "look-ahead bias") of the CP-Factors by rolling a 10-year estimation window forward. The period 01/1965 to 12/1974 serves as our initialization period. Afterwards, the CP-factor is estimated recursively. As a consequence, for example, the forward curve information from 01/1974 is only included in the estimation of the *CP*-Factor from 01/1974 to 01/1984.

Excess Returns. We obtain monthly series of excess returns of holding a bond with 12, 24, 36, 48, 60 and (smaller than) 120 months to maturity by taking the difference between the monthly return series from CRSP data and the return of a risk free asset (1 month T-bill) from Kenneth French's database. Quarterly return series and series at lower frequencies are constructed from these monthly time series.

A.3 Econometric Panel Approach

As mentioned in Section 4.2, we estimate panel regression with both endogenous and exogenous variables on the right-hand side of the equation, so that our general specification in Eq. (8) reads

$$e_{i,t} = a_1 e_{i,t-1} + \Xi_{i,t} \gamma + \Psi_t \delta + Z_t \beta + \epsilon_{i,t};$$

we refer to the main text for notation. We specify the error term as $\epsilon_{i,t} = \alpha_i + \mu_{i,t}$, which takes into account that forecasters' expectations may exhibit unobserved heterogeneity (in a fixed-effects panel regression setting).

As the unobserved component α_i is correlated with the lagged dependent variable, OLS would deliver inconsistent estimates. In the dynamic panel structure with a lagged dependent variable on the RHS, a fixed effects estimator is not appropriate either, as the differenced equation includes $\Delta e_{i,t-1} = e_{i,t-1} - e_{i,t-2}$ as a regressor, which is by construction correlated with the error term $\Delta \mu_t = \mu_{i,t} - \mu_{i,t-1}$. To circumvent these problems, Arellano and Bond (1991) propose the moment conditions with respect to the differenced equation

$$E[e_{i,t-s}(\mu_{i,t} - \mu_{i,t-1})] = 0$$
(A.1)

for $s \ge 2$ and t = 3, ..., T. To avoid potential problems caused by weak instruments and in order to improve efficiency, System GMM includes both the differenced as well as the additional orthogonality conditions for the errors in the level equation

$$E\left[\Delta e_{i,t-s}\epsilon_{i,t}\right] = 0 \tag{A.2}$$

for s = 1 and t = 4, ..., T (e.g., Blundell and Bond (1998)). System GMM Dynamic Panel Approaches are frequently applied to datasets in which the time series dimension T is small. As

our T is rather large (albeit smaller than the cross-sectional dimension N), the conventional System GMM approach generates too many instruments relative to N, which may cause Hansen's J statistics to underreject (Anatolyev and Gospodinov, 2010). To address this problem, we collapse the moment conditions shown in Equations (A.1) and (A.3) by addition into smaller subsets.²² Intuitively, this treats each moment condition to apply to all available periods instead of to each particular point in time individually, such that the moment conditions in Equation (A.1) are generated for $s \ge 2$ (instead of for $s \ge 2$ and t = 3, ..., T). As described by Cameron and Trivedi (2005, p. 765), we also construct the instruments from the *exogenous* variables Z_t and Ψ_t to

$$E\left[\Delta Z_{i,t}(\mu_{i,t} - \mu_{i,t-1})\right] = 0$$

$$E\left[Z_{i,t}\epsilon_{i,t}\right] = 0.$$
(A.3)

For the endogenous variables, we limit the collapsed System GMM-style instruments

$$E[X_{i,t-s}(\mu_{i,t} - \mu_{i,t-1})] = 0$$
(A.4)

to the second and third lags values (s = 2, 3) in the differenced equation, as well as

$$E\left[\Delta e_{i,t-1}\epsilon_{i,t}\right] = 0 \tag{A.5}$$

for the equation in levels. To rely on efficient estimates when errors exhibit heteroskedasticity, we report the results from two-step GMM. As this methodology may deliver downward biased standard errors in small samples, we apply the correction suggested by Windmeijer (2005) to obtain accurate inference.

A.4 Hodrick Standard Errors

We briefly review the construction of Hodrick (1992) standard errors used in our predictive regressions. Denote the vector of regression coefficients in Eq. (9) as $\phi_h = (\alpha_h \beta'_h)'$ and the RHS variables as $x_t = (1, z'_t)'$. The asymptotic distribution of ϕ_h when using GMM (Hansen, 1982) is $\sqrt{T}(\hat{\phi}_h - \phi_h) \sim \mathcal{N}(0, \Omega)$, where Ω is given by $\Omega = Z_0^{-1} S_0 Z_0^{-1}$ and $Z_0 = E(x_t x'_t)$. The idea of Hodrick's estimator is to exploit covariance stationarity and, hence, to sum the the explanatory variables into the past instead of summing residuals into the future. To this end, let

$$wk_t = e_{t+1}\left(\sum_{i=0}^{k-1} x_{t-i}\right)$$
 (A.6)

 $^{^{22}}$ For more details on this approach, see Roodman (2009).

where under the null hypothesis $\varepsilon_{t+h} = e_{t+1} + \ldots + e_{t+h}$, so that e_{t+1} denotes the one-step ahead forecast error. Estimates of e_{t+1} are obtained as the residual of a regression of returns on a constant. Finally, the spectral density S_0 is estimated as

$$\widehat{S}_0 = \frac{1}{T} \sum_{t=k}^T w k_t w k'_t \tag{A.7}$$

so that an estimate of Ω can be computed.

Table 1: Descriptive statistics: Risk premia

This table reports descriptive statistics for expected and realized changes in bond risk premia (Panel A and Panel B, respectively). Numbers in brackets are t-statistics for the means and are based on Newey-West HAC standard errors. AC(1) debotes first-order autocorrelation and numbers in parentheses are *p*-values for the test of no autocorrelation (based on the Ljung-Box Q statistic). *h* denotes the horizon and ranges from one to four quarters ahead.

		Expected	l changes	3		Realized	changes	
	h = 1	h = 2	h = 3	h = 4	h = 1	h = 2	h = 3	h = 4
Mean	-2.12	-2.03	-1.99	-2.28	-1.52	-3.07	-4.46	-5.83
t-stat.	[-3.27]	[-2.82]	[-2.48]	[-2.75]	[-2.79]	[-2.97]	[-3.13]	[-3.37]
Median	-1.75	-2.04	-1.96	-2.30	-1.72	-3.94	-4.49	-6.98
Stand. Dev.	0.95	1.05	1.17	1.20	1.17	1.84	2.28	2.66
Skewness	-0.64	-0.37	-0.26	-0.04	0.37	0.40	0.44	0.39
Kurtosis	3.75	3.48	3.53	3.48	2.77	2.48	2.83	2.78
AC(1)	0.69	0.69	0.69	0.69	0.21	0.57	0.73	0.73
p-value	(0.00)	(0.00)	(0.00)	(0.00)	(0.21)	(0.00)	(0.00)	(0.00)

Table 3: Relation of expected risk premium changes with standard bond factors

This table reports panel regression results where expected risk premium changes are regressed on standard yield curve factors (level, slope, and curvature) and the bond forecasting factor of Cochrane and Piazzesi (2005), denoted CP.

	(i)	(ii)	(iii)	(iv)	(v)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.425	0.358	0.317	0.432	0.321
,,	***(0.037)	***(0.040)	***(0.039)	***(0.035)	***(0.040)
Level	-0.098	, , , , , , , , , , , , , , , , , , ,	. ,	. ,	-0.101
	***(0.009)				***(0.009)
Slope		-0.072			-0.005
		***(0.006)			(0.008)
Curvature			0.060		0.058
			***(0.004)		***(0.003)
CP				-0.027	
				***(0.005)	
const.	0.501	0.039	0.070	-0.070	0.689
	***(0.047)	***(0.010)	***(0.011)	***(0.012)	***(0.051)
R_{COR}^2	0.33	0.38	0.43	0.32	0.47
J-Stat.	65.16	59.14	59.81	63.06	60.56
df	68	68	68	65	68
p-value	(0.58)	(0.77)	(0.75)	(0.55)	(0.73)
Test $\Delta \epsilon_t$ for AR(2)	0.006	-0.425	0.086	0.110	0.046
p-value	(1.00)	(0.67)	(0.93)	(0.91)	(0.96)
# Instr.	71	71	71	68	73
N	126	126	126	124	126
$N \cdot T$	$1,\!999$	$1,\!999$	$1,\!999$	$1,\!889$	1,999

Table 2: Determinants of expected bond risk premia

This table reports panel regression estimates where individual risk premium expectations are regressed on the lagged endogeneous variable $(E_{t-1,i}[\Delta \pi_{t-1+h}])$, and macro expectations: expected NGDP growth $(E_{t,i}[\Delta \text{NGDP}])$, expected RGDP growth $(E_{t,i}[\Delta \text{RGDP}])$ and expected inflation $(E_{t,i}[\Delta \text{INF}])$ over the next quarter. In addition, we present results for aggregate uncertainty about GDP growth ($\Psi(\text{RGDP}_{TY,NY})$) and inflation rates ($\Psi(\text{INF}_{TY,NY})$), where TY or NY denote uncertainty about this year or next year, respectively. $\Psi(INF_{NY})^{\perp}$ denotes inflation uncertainty orthogonalized with respect to GDP uncertainty. R_{COR}^2 denotes a Pseudo- R^2 , J denotes Hansen's J-statistic, "Test $\Delta \epsilon_t$ for AR(2)" shows the test for second-order residual autocorrelation, # Instr. denotes the total number of instruments used, N shows the total number of cross-sectional units, whereas NT denotes the total number of observations. Asterisks denote the level of significance, ***: 0.01, **: 0.05, *: 0.10.

	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)	(viii)	(ix)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.266	0.375	0.424	0.428	0.425	0.429	0.250	0.248	0.255
$E_{t,i}[\Delta { m RGDP}]$	0.512	(0.044)	(0.030)	(060.0)	(060.0)	(0.030)	(100.0) 0.444	(100.0)	(1.0.0)
	$^{***}(0.081)$						***(0.071)	***(0.067)	$^{***}(0.074)$
$E_{t,i}[\mathrm{INF}]$		0.135					0.051	0.039	0.044
		$^{***}(0.035)$					*(0.031)	(0.031)	(0.030)
$\Psi(\mathrm{RGDP}_{TY})$			-0.032 (0.047)						
$\Psi({\rm RGDP}_{NY})$				0.249 ***(0.055)			0.184 ***/0.055)		0.163 ***/0.055)
$\Psi(INF_{TV})$				(000.0)	0.167		(000.0)		(000.0)
					**(0.071)				
$\Psi(\mathrm{INF}_{NY})$					~	0.192		0.237	
-						(1.004)		(0,0,0)	
$\Psi({\rm INF}_{NY})^{\perp}$									0.178 ** (0.090)
const.	-0.431	-0.444	-0.092	-0.090	-0.094	-0.092	-0.517	-0.515	-0.510
	$^{***}(0.057)$	***(0.091)	$^{***}(0.012)$	***(0.012)	$^{***}(0.012)$	$^{***}(0.012)$	$^{***}(0.091)$	$^{***}(0.092)$	$^{***}(0.092)$
R^2_{COR}	0.36	0.29	0.34	0.34	0.34	0.34	0.37	0.37	0.37
J-Stat.	64.85	62.63	67.46	62.27	63.83	63.26	64.66	65.38	65.66
df	20	20	68	68	68	68	72	72	72
p-value	(0.65)	(0.72)	(0.50)	(0.67)	(0.62)	(0.64)	(0.72)	(0.70)	(0.69)
Test $\Delta \epsilon_t$ for AR(2)	1.463	0.656	-0.011	0.571	0.079	0.18	1.861	1.777	1.897
p-value	(0.14)	(0.51)	(0.99)	(0.57)	(0.94)	(0.86)	*(0.06)	*(0.08)	*(0.06)
# Instr.	73	73	71	71	71	71	22	22	78
N	116	114	126	126	126	126	114	114	114
$N \cdot T$	1,639	1,575	1,999	1,999	1,999	1,999	1,553	1,553	1,553

 Table 4: Combining macro expectations, uncertainty and bond factors

This table reports panel regression results where expected risk premium changes are regressed on standard yield curve factors (level, slope, and curvature) and the bond forecasting factor of Cochrane and Piazzesi (2005), denoted CP.

	(i)	(ii)	(iii)	(iv)	(v)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.264	0.207	0.228	0.301	0.234
	***(0.047)	***(0.049)	***(0.047)	***(0.049)	***(0.049)
$E_{t,i}[\Delta \text{RGDP}]$	0.444	0.428	0.300	0.393	0.313
,	***(0.073)	***(0.072)	***(0.082)	***(0.070)	***(0.083)
$E_{t,i}[INF]$	0.071	0.025	-0.009	0.031	0.019
, <u> </u>	**(0.030)	(0.029)	(0.029)	(0.037)	(0.029)
$\Psi(\mathrm{RGDP}_{NY})$	0.095	0.083	0.102	0.110	0.045
	(0.058)	(0.058)	*(0.052)	***(0.058)	**(0.050)
$\Psi(\mathrm{INF}_{NY})^{\perp}$	-0.014	0.076	0.306	0.146	0.096
	(0.087)	(0.086)	***(0.085)	*(0.085)	(0.084)
Level	-0.117				-0.083
	***(0.001)				***(0.013)
Slope		-0.077			-0.025
		***(0.007)			**(0.011)
Curvature			0.051		0.037
			***(0.006)		***(0.008)
CP				-0.042	
				***(0.006)	
const.	0.134	-0.308	-0.136	-0.497***	0.295
	(0.081)	***(0.080)	(0.116)	(0.089)	***(0.109)
R_{COR}^2	0.43	0.47	0.45	0.34	0.51
J-Stat.	66.34	65.54	62.82	58.73	61.72
df	72	72	72	69	72
p-value	(0.67)	(0.69)	(0.77)	(0.81)	(0.80)
Test $\Delta \epsilon_t$ for AR(2)	1.524	1.469	1.034	1.125	0.943
p-value	(0.13)	(0.14)	(0.30)	(0.26)	(0.35)
# Instr.	79	79	79	76	81
N	114	114	114	111	114
$N \cdot T$	$1,\!553$	1,553	1,553	1,465	1,553

Table 5: Combining macro expectations, uncertainty, and proxies for risk

This table reports panel regression results where expected risk premium changes are regressed on
lagged expectations, macro expectations, aggregate uncertainty measures as well as risk-related
variables. For the latter, we include lagged T-bond and/or T-bill volatility (based on realized
volatility obtained from adding squared changes on a daily frequency, denoted $\sigma(\text{TBOND})$ and
σ (TBILL)), and the yield spread between BAA and AAA rated bonds.

	(i)	(ii)	(iii)	(iv)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.263	0.288	0.217	0.264
	***(0.047)	***(0.051)	***(0.049)	***(0.047)
$E_{t,i}[\Delta \text{RGDP}]$	0.444	0.345	0.289	0.155
	***(0.075)	***(0.078)	***(0.087)	*(0.089)
$E_{t,i}[INF]$	0.017	0.057	-0.005	-0.020
	(0.028)	*(0.029)	(0.033)	(0.030)
$\Psi(\mathrm{RGDP}_{NY})$	0.070	0.118	0.165	0.054
	(0.054)	**(0.048)	***(0.051)	(0.049)
$\Psi(\text{INF}_{TY})$	0.222	0.177	0.167	0.203
	***(0.084)	**(0.089)	**(0.083)	**(0.079)
σ (TBOND)	-0.859			-0.727
	***(0.139)			***(0.140)
σ (TBILL)		-0.480		-0.469
		***(0.099)		***(0.093)
BAA-AAA			-0.218	-0.200
			***(0.046)	***(0.040)
const.	-0.290	-0.401	-0.076	0.224
	***(0.0967)	***(0.095)	(0.152)	(0.143)
R^2_{COB}	0.40	0.41	0.40	0.45
J-Stat.	69.39	66.22	66.66	67.08
df	72	72	72	72
p-value	(0.57)	(0.67)	(0.66)	(0.64)
Test $\Delta \epsilon_t$ for AR(2)	2.029	2.004	1.347	1.328
p-value	(0.04)	(0.05)	(0.18)	(0.18)
# Instr.	79	79	79	81
N	114	114	114	114
$N \cdot T$	1,553	1,553	1,553	1,553

Table 6: Combining macro expectations, uncertainty, and real-time macro factors

This table reports panel regression results where expected risk premium changes are regressed on lagged macro expectations, macro uncertainty, and real-time macro factors. As additional macro factors we consider growth rates in CPI inflation (Δ CPI), Industrial Production (Δ IP), GDP (Δ GDP), and M2 (Δ M2).

	(i)	(ii)	(iii)	(iv)	(v)
\overline{E} [A –]	0.962	0.911	0.997	0.240	0.917
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.203	0.211 ***(0.040)	0.227 ***(0.049)	0.249 ***(0.049)	0.217
	0.451	0.411	0.422	0.460	(0.047)
$E_{t,i}[\Delta \mathrm{KGDP}]$	0.431 ***(0.072)	0.411 ***(0.079)	0.433 ***(0.076)	0.400 ***(0.079)	0.308
	(0.073)	0.014	(0.076)	0.078)	(0.079)
$E_{t,i}[INF]$	0.062	0.014	(0.015)	0.040	(0.033)
	$^{*}(0.031)$	(0.035)	(0.032)	(0.035)	(0.033)
$\Psi(\text{RGDP}_{NY})$	0.121	0.151	0.107	0.147	0.115
> I	**(0.058)	***(0.053)	(0.057)	(0.075)	**(0.051)
$\Psi(INF_{NY})^{\perp}$	0.229	0.129	0.190	0.155	0.213
	**(0.102)	(0.091)	**(0.087)	(0.106)	**(0.097)
ΔCPI	-0.023				-0.042
	(0.011)				*(0.011)
ΔIP		0.021			0.025
		***(0.005)			***(0.004)
ΔGDP			5.341		
			***(1.166)		
$\Delta M2$				-0.002	
				(0.006)	
const.	-0.487	-0.455	-0.584	-0.509	-0.369
	***(0.089)	***(0.100)	***(0.095)	***(0.115)	***(0.094)
R^2_{COR}	0.37	0.39	0.40	0.37	0.41
J-Stat.	62.4	66.19	67.30	68.71	62.71
df	72	72	72	72	72
<i>p</i> -value	(0.78)	(0.67)	(0.64)	(0.59)	(0.77)
Test $\Delta \epsilon_t$ for AR(2)	1.837	1.390	1.560	1.856	1.258
p-value	(0.07)	(0.17)	(0.12)	(0.06)	(0.21)
# Instr.	79	79	79	79	80
N	114	114	114	114	114
$N \cdot T$	1,553	1,553	1,553	1,553	1,553

Table 7: Predictive Regressions

This table reports predictive regressions of future bond excess returns rx(m) on our proxy for expected term premium changes $E_t[\Delta \pi_{t+h}]$. Excess returns are based on CRSP bond returns for maturities of $m = 1, \ldots, 5, 10$ years and the average return over all maturities minus the return to holding a three month T-bill. Panels A – D show results for forecast horizons h of one to four quarters and we match forecast horizons with the horizon of our proxy for expected term premia. We report t-statistics based on Newey-West (1987) standard errors (t_{NW}) as well as based on Hodrick (1992) standard errors (t_H) .

	rx(avg)	rx(1Y)	rx(2Y)	rx(3Y)	rx(4Y)	rx(5Y)	rx(10Y)
	Р	anel A: Fo	precast ho	rizon $h =$	1 quarter		
const.	0.25	0.09	0.14	0.21	0.29	0.33	0.46
t_{NW}	[1.19]	[2.63]	[1.27]	[1.12]	[1.14]	[1.05]	[1.20]
t_H	[1.33]	[3.01]	[1.43]	[1.25]	[1.26]	[1.15]	[1.32]
$E_t[\Delta \pi_{t+1}]$	-1.09	-0.19	-0.77	-1.20	-1.47	-1.60	-1.33
t_{NW}	[-1.38]	[-1.53]	[-1.72]	[-1.60]	[-1.50]	[-1.36]	[-1.00]
t_H	[-1.32]	[-1.53]	[-1.65]	[-1.52]	[-1.42]	[-1.27]	[-0.98]
R^2	0.02	0.04	0.04	0.03	0.02	0.01	0.00
	Р	anel B: Fo	recast hor	izon $h = 2$	2 quarters		
const.	0.53	0.17	0.31	0.48	0.63	0.69	0.91
t_{NW}	[1.54]	[3.02]	[1.66]	[1.52]	[1.51]	[1.37]	[1.47]
t_H	[1.52]	[3.40]	[1.75]	[1.54]	[1.48]	[1.30]	[1.42]
$E_t[\Delta \pi_{t+1}]$	-1.96	-0.38	-1.29	-1.95	-2.53	-2.91	-2.70
t_{NW}	[-1.91]	[-2.22]	[-2.18]	[-2.02]	[-1.99]	[-1.90]	[-1.52]
t_H	[-1.57]	[-2.06]	[-1.94]	[-1.69]	[-1.62]	[-1.50]	[-1.28]
R^2	0.05	0.10	0.08	0.06	0.06	0.05	0.02
	Ра	anel C: Fo	recast hor	izon $h = 3$	3 quarters		
const.	0.82	0.26	0.49	0.75	0.98	1.07	1.35
t_{NW}	[1.76]	[3.06]	[1.74]	[1.67]	[1.72]	[1.59]	[1.70]
t_H	[1.60]	[3.58]	[1.86]	[1.65]	[1.57]	[1.38]	[1.45]
$E_t[\Delta \pi_{t+1}]$	-2.31	-0.49	-1.51	-2.20	-2.87	-3.34	-3.43
t_{NW}	[-1.98]	[-2.57]	[-2.26]	[-1.98]	[-1.98]	[-1.93]	[-1.71]
t_H	[-1.59]	[-2.25]	[-2.00]	[-1.69]	[-1.61]	[-1.51]	[-1.35]
R^2	0.06	0.13	0.09	0.06	0.06	0.05	0.04
	Pa	anel D: Fo	recast hor	izon $h = 4$	4 quarters		
const.	0.86	0.31	0.53	0.80	1.04	1.11	1.38
t_{NW}	[1.62]	[2.93]	[1.51]	[1.47]	[1.54]	[1.42]	[1.64]
t_H	[1.24]	[3.09]	[1.49]	[1.29]	[1.22]	[1.04]	[1.10]
$E_t[\Delta \pi_{t+1}]$	-3.63	-0.73	-2.22	-3.36	-4.43	-5.23	-5.80
t_{NW}	[-3.17]	[-3.65]	[-3.17]	[-2.95]	[-3.07]	[-3.11]	[-3.10]
t_H	[-2.05]	[-2.68]	[-2.39]	[-2.13]	[-2.06]	[-1.97]	[-1.82]
R^2	0.13	0.22	0.15	0.12	0.12	0.12	0.11



(a) Time-series of RGDP and Inflation uncertainty



(b) Correlation of RGDP and Inflation uncertainty

Figure 1: Macro uncertainty

This figure shows plots of aggregate uncertainty about the following year's real gdp growth (blue, dashed line) and next year's inflation (red, solid line) in Panel (a) and cross-plots of the two series in Panel (b). ρ denotes the simple linear correlation coefficient between the two series.



Figure 2: Expected and realized changes in term premia

This figure shows plots of expected (red, solid line) and realized changes (blue, dashed line) in term premia over different horizons of h = 1, 2, 3, and 4 quarters.



Figure 3: Bond yield factors

This figure shows time-series plots of level, slope, and curvature based on Diebold and Li (2006) and the bond forecasting factor of Cochrane and Piazzesi (2005). Note that we have multiplied the slope factor with minus one so that higher readings of "slope" correspond to a steeper yield curve. The scaling of the graphs is in basis points.

Supplementary Appendix to accompany

Macro Expectations, Aggregate Uncertainty,

and Expected Term Premia

	$E_t[\Delta \pi_{t+1}]$	$E_t[\Delta \mathrm{RGDP}]$	$E_t[\Delta \text{INF}]$	ΔCPI	$\Delta RGDP$	ΔIP	$\Delta M2$	Level
$E_t[\Delta \pi_{t+1}]$	1.00							
$E_t[\Delta \mathrm{RGDP}]$	0.55	1.00						
$E_t[\Delta INF]$	0.34	0.32	1.00					
ΔCPI	0.16	0.08	0.31	1.00				
$\Delta \mathrm{RGDP}$	0.57	0.43	0.12	0.27	1.00			
ΔIP	0.66	0.59	0.54	0.35	0.71	1.00		
$\Delta \ \mathrm{M2}$	-0.30	-0.33	-0.70	-0.63	-0.26	-0.62	1.00	
Level	-0.35	0.05	0.51	0.08	-0.26	0.11	-0.47	1.00
Slope	-0.63	-0.03	-0.10	-0.16	-0.56	-0.44	-0.01	0.51
Curvature	0.67	0.46	0.56	0.07	0.39	0.59	-0.23	0.09
CP	-0.33	0.27	0.24	0.14	-0.20	0.08	-0.32	0.56
$\sigma(\mathrm{TBOND})$	-0.34	-0.48	-0.09	0.36	-0.05	-0.17	-0.02	0.02
$\sigma(\mathrm{TBILL})$	-0.40	-0.21	-0.27	-0.19	-0.18	-0.26	0.16	0.17
BAA-AAA	-0.63	-0.69	-0.62	-0.21	-0.49	-0.78	0.44	-0.21
$\Psi({ m RGDP}_{NY})$	0.09	0.15	-0.16	-0.14	0.19	0.00	0.28	-0.14
$\Psi(\mathrm{INF}_{NY})$	0.07	-0.02	-0.07	0.18	-0.06	-0.03	0.02	-0.25
$\Psi(\mathrm{INF}_{NY})^{\perp}$	0.03	-0.12	0.02	0.30	-0.19	-0.04	-0.16	-0.20
	Slope	Curvature	CP	$\sigma(\text{TBOND})$	$\sigma(\text{TBILL})$	BAA-AAA	$\Psi({\rm RGDP}_{NY})$	$\Psi({\rm INF}_{NY})$
Slope	1.00							
Curvature	-0.54	1.00						
CP	0.49	-0.14	1.00					
$\sigma(\mathrm{TBOND})$	0.01	-0.25	0.12	1.00				
$\sigma(\mathrm{TBILL})$	0.32	-0.30	0.01	0.13	1.00			
BAA-AAA	0.30	-0.72	-0.03	0.19	0.32	1.00		
$\Psi(\mathrm{RGDP}_{NY})$	-0.13	0.06	-0.24	-0.08	-0.18	-0.02	1.00	
$\Psi(\mathrm{INF}_{NY})$	-0.10	-0.14	-0.25	0.05	-0.10	0.12	0.54	1.00
$\Psi(\mathrm{INF}_{NY})^{\perp}$	-0.04	-0.21	-0.14	0.11	0.01	0.16	-0.01	0.84

Table A.I: Correlation coefficients

This table shows a

Table A.II: Determinants of expected bond risk premia

The setup of this table is identical to table 2, but we include an NBER recession dummy.

	(i)	(ii)	(iii)	(iv)	(v)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.332	0.225	0.235	0.233	0.241
	$^{***}(0.040)$	$^{***}(0.053)$	***(0.053)	$^{***}(0.052)$	$^{***}(0.051)$
$E_{t,i}[\Delta \mathrm{RGDP}]$		0.428	0.411	0.449	0.425
-		***(0.074)	***(0.078)	***(0.078)	***(0.084)
$E_{t,i}[\mathrm{INF}]$		0.042	0.047	0.036	0.041
		(0.031)	(0.031)	(0.031)	(0.031)
$\Psi(\mathrm{RGDP}_{NY})$			0.162		0.146
×.			$^{***}(0.054)$		***(0.054)
$\Psi(\mathrm{INF}_{NY})$				0.213	
~				***(0.070)	
$\Psi(\mathrm{INF}_{NY})^{\perp}$					0.165
					*(0.090)
Recession	-0.273	-0.090	-0.079	-0.071	-0.070
	***(0.029)	$^{**}(0.042)$	*(0.041)	*(0.042)	*(0.042)
const.	-0.071	-0.479	-0.478	-0.480	-0.47
	$^{***}(0.012)$	***(0.099)	***(0.098)	$^{***}(0.100)$	***(0.099
R^2_{COR}	0.38	0.37	0.37	0.37	0.38
J - Stat.	62.04	65.85	65.4	65.99	66.6
df	68	72	72	72	:2
p-value	(0.68)	(0.68)	(0.70)	(0.68)	(0.66)
Test $\Delta \epsilon_t$ for AR(2)	-1.578	0.992	1.432	1.421	1.53(
p-value	(0.12)	(0.32)	(0.15)	(0.16)	(0.13)
# Instr.	71	22	78	78	52
N	126	114	114	114	11^{-1}
$N \cdot T$	1,999	1,553	1,553	1,553	1.553

	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(vii)	(viii)	(ix)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.343 ***(0.064)	0.419 ***(0.059)	0.443 ***(0.045)	0.441 ***/0.045)	0.452 ***(0.045)	0.452 ***(0.044)	0.346	0.353 ***(0.067)	0.357 ***(0.066)
$E_{t,i}[\Delta { m RGDP}]$	0.165^{***}	(000.0)	(010.0)	(0+0.0)	(010.0)	(===0.0)	0.137	0.164	0.145
3	(0.053)						**(0.054)	$^{***}(0.054)$	$^{**}(0.056)$
$E_{t,i}[\mathrm{INF}]$		0.113					0.019	-0.001	0.011
		(0.072)					(0.060)	(0.060)	(0.060)
$\Psi({ m RGDP}_{TY})$			0.098 (0.084)						
$\Psi({\rm RGDP}_{NY})$			~	0.394			0.360		0.326
				(00.009)			(0.053)		(0.053)
$\Psi(\mathrm{INF}_{TY})$					0.406 *** (0.113)				
$\Psi(\mathrm{INF}_{NY})$						0.399		0.410	
						(0.091)		$^{}(0.111)$	
$\Psi({\rm INF}_{NY})^{\perp}$									0.267 *(0.141)
const.	-0.546	-0.391	-0.094	-0.094	-0.095	-0.094	-0.521	-0.543	-0.522
	***(0.143)	**(0.197)	***(0.018)	***(0.018)	***(0.018)	***(0.018)	**(0.207)	**(0.211)	**(0.212)
R^2_{COR}	0.47	0.45	0.48	0.48	0.47	0.48	0.49	0.48	0.49
J - Stat.	61.11	62.36	64.05	61.76	62.97	63.05	62.86	62.36	63.02
df	20	20	68	68	68	68	72	72	72
p-value	(0.77)	(0.73)	(0.61)	(0.69)	(0.65)	(0.65)	(0.77)	(0.78)	(0.77)
Test $\Delta \epsilon_t$ for AR(2)	0.885	0.347	0.926	1.316	0.903	1.064	1.241	1.137	1.335
p-value	(0.38)	(0.73)	(0.35)	(0.19)	(0.37)	(0.29)	(0.21)	(0.26)	(0.18)
# Instr.	73	73	71	71	71	71	22	22	78
N	113	112	124	124	124	124	111	111	111
$N \cdot T$	1,471	1,432	1,847	1,847	1,847	1,847	1,409	1,409	1,409

 Table A.III: Determinants of expected bond risk premia

The setup of this table is identical to table 2 but here we investigate expectations for four quarters ahead, i.e. h = 4.

Table A.IV: Relation of expected risk premium changes with risk-related factors

This table reports panel regression results where expected risk premium changes are regressed on lagged expectations, lagged yield changes, and risk-related variables. We include lagged changes in T-bond and/or T-bill yields (Δ TBOND, Δ TBILL) over the last quarter, lagged T-bond and/or T-bill volatility (based on realized volatility obtained from summing up squared changes on a daily frequency, denoted σ (TBOND) and σ (TBILL)), and the yield spread between BAA and AAA rated bonds.

	(i)	(ii)	(iii)	(iv)	(v)	(vi)
$E_{t-1,i}[\Delta \pi_{t-1+h}]$	0.453	0.463	0.414	0.425	0.288	0.384
,	***(0.036)	***(0.038)	***(0.033)	***(0.037)	***(0.040)	***(0.041)
ΔTBOND	-0.696	· · · ·	· · · ·	· · · ·	× ,	-0.262
	***(0.142)					*(0.151)
$\Delta \mathrm{TBILL}$		0.279				0.148
		***(0.019)				***(0.027)
$\sigma(\text{TBOND})$		· · · ·	-1.128			-0.707
			***(0.120)			***(0.103)
$\sigma(\text{TBILL})$			· · · ·	-0.850		-0.351
· · · · ·				***(0.066)		***(0.072)
BAA-AAA				· · · ·	-0.309	-0.151
					***(0.021)	***(0.025)
const.	-0.093	-0.065	0.092	0.035	0.186	0.228
	***(0.011)	***(0.011)	***(0.023)	**(0.015)	***(0.020)	***(0.027)
R_{COR}^2	0.35	0.47	0.38	0.41	0.39	0.50
J-Stat.	63.02	58.96	64.76	63.79	68.11	66.74
df	68	68	68	68	68	68
p-value	(0.65)	(0.78)	(0.59)	(0.62)	(0.47)	(0.52)
Test $\Delta \epsilon_t$ for AR(2)	-0.402	0.718	1.113	0.850	-0.018	1.258
<i>p</i> -value	(0.69)	(0.47)	(0.27)	(0.40)	(0.99)	(0.21)
# Instr.	71	71	71	71	71	75
N	126	126	126	126	126	126
$N \cdot T$	1,999	1,999	$1,\!999$	$1,\!999$	$1,\!999$	1,999

	rx(avg)	rx(1Y)	rx(2Y)	rx(3Y)	rx(4Y)	rx(5Y)	rx(10Y)				
Panel A: Forecast horizon $h = 1$ quarter											
const.	0.48	0.12	0.28	0.44	0.58	0.66	0.77				
t_{NW}	[2.84]	[4.64]	[2.98]	[2.76]	[2.76]	[2.59]	[2.75]				
t_H	[2.55]	[4.80]	[2.92]	[2.58]	[2.49]	[2.27]	[2.39]				
CP	0.00	0.00	0.00	0.00	0.00	0.01	0.01				
t_{NW}	[4.74]	[2.62]	[3.51]	[4.18]	[4.60]	[4.83]	[5.34]				
t_H	[3.08]	[2.36]	[2.80]	[2.93]	[3.02]	[3.04]	[3.21]				
R^2	0.19	0.10	0.14	0.16	0.18	0.19	0.21				
		Panel B:	Forecast h	orizon h =	= 2 quarte	ers					
const.	0.90	0.24	0.53	0.83	1.09	1.23	1.46				
t_{NW}	[2.84]	[4.42]	[2.83]	[2.68]	[2.74]	[2.63]	[2.84]				
t_H	[2.47]	[4.81]	[2.87]	[2.51]	[2.42]	[2.19]	[2.31]				
CP	0.00	0.00	0.00	0.00	0.01	0.01	0.01				
t_{NW}	[4.05]	[1.93]	[2.76]	[3.46]	[3.97]	[4.36]	[4.59]				
t_H	[2.50]	[1.70]	[2.18]	[2.31]	[2.41]	[2.45]	[2.73]				
R^2	0.17	0.06	0.10	0.13	0.15	0.17	0.21				
Panel C: Forecast horizon $h = 3$ quarters											
const.	1.25	0.34	0.75	1.15	1.51	1.70	2.03				
t_{NW}	[2.62]	[4.07]	[2.54]	[2.42]	[2.50]	[2.41]	[2.69]				
t_H	[2.41]	[4.95]	[2.84]	[2.45]	[2.35]	[2.11]	[2.23]				
CP	0.01	0.00	0.00	0.00	0.01	0.01	0.01				
t_{NW}	[2.57]	[1.58]	[1.89]	[2.18]	[2.40]	[2.60]	[3.13]				
t_H	[2.34]	[1.79]	[2.11]	[2.18]	[2.23]	[2.23]	[2.57]				
R^2	0.14	0.06	0.09	0.11	0.13	0.14	0.19				
		Panel D:	Forecast h	orizon h	= 4 quarte	Y) $rx(5Y)$ $rx(10Y)$ narter 58 0.66 0.77 76] [2.59] [2.75] 49] [2.27] [2.39] 00 0.01 0.01 50] [4.83] [5.34] 52] [3.04] [3.21] 18 0.19 0.21 arters 09 1.23 1.46 74] [2.63] [2.84] 42] [2.19] [2.31] 01 0.01 0.01 97 [4.36] [4.59] 41] [2.45] [2.73] 15 0.17 0.21 arters 51 1.70 2.03 50] [2.41] [2.69] 35] [2.11] [2.23] 01 0.01 0.01 40] [2.60] [3.13] 23] [2.23] [2.57] 13 0.14 0.19 arters 97 2.22 2.64 41] [2.35]					
const.	1.63	0.45	0.99	1.51	1.97	2.22	2.64				
t_{NW}	[2.54]	[3.83]	[2.39]	[2.31]	[2.41]	[2.35]	[2.71]				
t_H	[2.39]	[4.97]	[2.83]	[2.43]	[2.33]	[2.10]	[2.19]				
CP	0.01	0.00	0.00	0.00	0.01	0.01	0.01				
t_{NW}	[1.87]	[1.38]	[1.45]	[1.57]	[1.70]	[1.82]	[2.35]				
t_H	[1.90]	[1.70]	[1.76]	[1.75]	[1.78]	[1.79]	[2.11]				
R^2	0.09	0.04	0.05	0.06	0.08	0.09	0.14				

Table A.V: Predictive Regressions with the CP factor

This table reports predictive regressions of future bond excess returns rx(m) on the bond forecasting factor by Cochrane and Piazzesi (2005). The setup is identical to that of Table 7.